# **Guide to the Aarhus Convention Compliance Committee**





# Guide to the Aarhus Convention Compliance Committee



#### NOTE

Symbols of United Nations documents are composed of capital letters combined with figures. Mention of such a symbol indicates a reference to a United Nations document.

The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of the Secretariat of the United Nations concerning the legal status of any country, territory, city or area, or of its authorities, or concerning the delimitation of its frontiers or boundaries.

This publication may be reproduced in whole or in parts and in any form for educational or non-profit purposes, provided acknowledgement of the source is made. The UNECE would appreciate receiving a copy of any publication that uses this publication as a source.

Convention on Access to Information, Public Participation in Decision-making and Access to Justice in Environmental Matters (Aarhus Convention)
Environment Division
United Nations Economic Commission for Europe
Palais des Nations
CH-1211 Geneva 10
Switzerland

Phone: +41 22 917 4226

Email: aarhus.compliance@un.org

Website: http://www.unece.org/env/pp/welcome.html

#### **Preface**

This Guide is intended to explain the functions and working methods of the Aarhus Convention Compliance Committee for Parties, communicants and observers that may engage with the Committee's procedures. The Committee's working methods are framed by decision I/7 of the Meeting of the Parties (adopted at its first session in Lucca, October 2002), which established the Committee.

This is the second edition of the Guide. The first edition, entitled the Guidance Document on the Aarhus Convention Compliance Mechanism, was adopted in 2009.

In the decade since then, in order to respond to the needs of its evolving caseload and to improve the efficiency and effectiveness of its work, while at all times ensuring fairness and due process, the working methods of the Committee have developed considerably. The second edition of the Guide provides a clear and easy-to-follow explanation of how the Committee carries out each aspect of its work.

The second edition of the Guide was prepared through a thorough, transparent and participatory process between the Committee's 47<sup>th</sup> meeting (16-19 December 2014) and the 63<sup>rd</sup> meeting (11-15 March 2019). During this period, six drafts were prepared, with each draft published on the Committee's website prior to the Committee meeting at which it was to be discussed. At each meeting, Parties and observers had the opportunity to comment on the text and also to send written comments to the secretariat after the meeting. The comments received on each draft were then taken into account in the preparation of the subsequent draft. The Guide was adopted at the Committee's 63<sup>rd</sup> meeting.

On behalf of the Committee, I hope that the Guide will be a user-friendly tool to facilitate the engagement of Parties and members of the public in the Committee's work.

Jonas Ebbesson

Chair of the Compliance Committee

### Contents<sup>1</sup>

I.	Introduction	
Prov Repo Moni Meas	Functions and powers of the Committee sider any submission, referral, communication or request ide advice and facilitate assistance to individual Parties ort to the Meeting of the Parties itor reporting requirements sures by the Meeting of the Parties to address compliance issues -sessional measures by the Committee to address compliance issues	8 8 8 9 9
Oper NGC Use Work Tran Use Publ Clos Virtu Deci Deci Miss	Committee's working methods ess to information about cases before the Committee in sessions with public participation Os and the Committee of audio-conferencing and video-conferencing king language slation of hyperlinks ication of meetings and documentation ed sessions al meetings sion-making sion-making by e-mail ions of the secretariat	11 12 12 13 13 14 14 15 16 16
	Composition and election of the Committee sposition of the Committee tion of Committee members Nomination Election Geographic representation Rotation	18 18 18 18 18 19
Deci The	Responsibilities of Committee members artiality and conscientiousness sion-making by consensus role of curator flict of interest	20 20 20 20 20
Dete Resp Infor Whe Prep	Overview of compliance review – step by step gering compliance review ermination of preliminary admissibility conse by the Party concerned mation gathering are needed a hearing with the parties concerned earation of findings ow-up if non-compliance found	23 23 23 23 23 24 24 24

<sup>&</sup>lt;sup>1</sup> The Guide should be considered to be a living document and may be subject to further development.

VII. Triggering review of compliance Types of compliance cases	25 25
a. Communications from the public	26
Upon receipt of a new communication	26
Decision by Chair and Vice-Chair(s) to forward communication to Committee for review  Determination of admissibility Use of domestic remedies Summary proceedings Response by the Party concerned b. Submissions by Parties concerning other Parties Upon receipt of a new submission	26 27 30 31 32 34 34
Response from the Party concerned	35
c. Submissions by Parties concerning own compliance	36
d. Referrals by the secretariat e. Requests by the Meeting of the Parties	37 39
VIII. Common elements for all compliance review procedures Information gathering Comments and information submitted during the proceeding Where needed, a hearing with the parties concerned Preparation of draft findings Adoption of findings and recommendations	40 40 42 43 45 46
Recommendations directly to the Party concerned Consideration by the Meeting of the Parties Committee's review of the implementation of decision of the Meeting of the Parties Report to the Meeting of the Parties on the implementation of its decision on compliance Any developments subsequent to the finalization of the Committee's report to the Meeting of the Parties Issuance of a caution	47 47 48 49 50
Summary of key points Who can submit a communication Specify the Party concerned by the communication Timing of a communication and of the related facts Form of the communication Presentation of the facts of alleged non-compliance The nature of alleged non-compliance Provisions of the Convention relating to the alleged non-compliance Use of domestic remedies Use of other international procedures Confidentiality Supporting documentation Language of the communication and related documentation To whom communications should be addressed and how	52 52 53 54 55 55 56 56 56 57 58
Annex 1: Format for communications and submissions	59

#### I. Introduction

- 1. Since the 1990s, there has been a growing trend in international treaty law-making to develop mechanisms that facilitate, promote and enforce compliance with the commitments undertaken by the Parties. Article 15 of the Convention on Access to Information, Public Participation in Decision-making and Access to Justice in Environmental Matters (the Aarhus Convention) requires the Parties to set up arrangements of a non-confrontational, non-judicial and consultative nature to review compliance with the Convention; such arrangements are required to allow for public involvement and may include the option of considering communications from members of the public on matters related to the Convention.
- 2. On the basis of this provision, at its first session (Lucca, October 2002) the Meeting of the Parties adopted decision I/7 on the review of compliance. Decision I/7 establishes an advanced compliance mechanism for the Convention: It creates the Compliance Committee as the main body for the review of compliance and sets out the structure and functions of this body as well as the procedures to review compliance.
- 3. The Committee reviews Parties' compliance with the provisions of the Convention and reports to the Meeting of the Parties.
- 4. A review of a specific Party's compliance may be triggered in five ways:
  - (a) A Party may make a submission about compliance by another Party;
  - (b) A Party may make a submission concerning its own compliance:
  - (c) The secretariat may make a referral to the Committee;
  - (d) Members of the public may make communications concerning a Party's compliance with the Convention;<sup>2</sup>
  - (e) The Meeting of the Parties may request the Committee to examine a Party's compliance with the Convention.<sup>3</sup>
- 5. In addition, a Party may make a request for advice or assistance.4
- 6. The compliance mechanism of the Aarhus Convention is one of the few such mechanisms in international environmental law which allows members of the public to communicate their concerns about a Party's compliance directly to a board of independent experts, the Compliance Committee, which has the mandate to examine the merits of the case. The Committee adopts findings and if non-compliance is found, may make recommendations either to the Meeting of the Parties, or, with the Party's agreement, directly to the Party concerned.
- 7. The Committee may also examine compliance issues on its own initiative<sup>5</sup> and

<sup>&</sup>lt;sup>2</sup> The triggers in paragraphs (a)-(d) are described in paras. 15-18 of the annex to decision I/7.

<sup>&</sup>lt;sup>3</sup> This trigger is not expressly referred to in the annex to decision I/7, but it follows from the Meeting of the Parties' general decision-making capacity.

<sup>&</sup>lt;sup>4</sup> Annex to decision I/7, para. 37(a).

<sup>&</sup>lt;sup>5</sup> Annex to decision I/7, para. 14.

make recommendations; prepare reports on compliance with or implementation of the provisions of the Convention at the request of the Meeting of the Parties; provide advice or facilitate assistance to individual Parties regarding the implementation of the Convention; and monitor, assess and facilitate the implementation of and compliance with the reporting requirements under article 10(2) of the Convention.

- 8. This Guide is intended to explain the functions of the Compliance Committee and its working methods for Parties, communicants and observers that may engage with the Committee's procedures. It is divided into ten sections and one annex. Following the introduction to the Committee in the present section, section II set outs the Committee's functions and powers and section III describes the Committee's working methods. Section IV explains the composition of the Committee and the procedure through which Committee members are elected and section V sets out the responsibilities of Committee members. Section VI provides a step-by-step overview of the compliance review procedure and section VIII explains how the Committee's review of compliance may be triggered. Section VIII describes the elements of the Committee's procedure for reviewing compliance which apply to all the types of cases in its caseload. Section IX explains the procedure for following-up on a finding that a Party is in non-compliance. Finally, section X highlights some useful information for members of the public when preparing a communication to the Committee and the annex contains the required format for a communication.
- 9. The information contained in this guidance document is primarily based on:
  - (a) The Convention, notably article 15;
  - (b) Decisions I/7, II/5, III/6, IV/9, V/9 and VI/8 of the Meeting of the Parties;
  - (c) The reports of the Committee's meetings.
- 10. All relevant documentation concerning the Compliance Committee is available at http://www.unece.org/env/pp/cc.html.
- 11. Further information concerning the Convention is available on the Convention's website: www.unece.org/env/pp/welcome.html.

## II. Functions and powers of the Committee

# Consider any submission, referral, communication or request in order to examine compliance and implementation

- 12. In accordance with paragraph 13 of the annex to decision I/7, the Committee:
  - (a) Considers any submission, referral or communication concerning compliance by Parties with their obligations under the Convention;
  - (b) Prepares, at the request of the Meeting of the Parties, a report on compliance with or implementation of the provisions of the Convention;
  - (c) Monitors, assesses and facilitates the implementation of and compliance with the reporting requirements under article 10(2) of the Convention.
- 13. The Committee may examine compliance issues and make recommendations if and as appropriate.<sup>6</sup>
- 14. The Committee's role is to further the implementation of the Convention. It is not a redress mechanism.

#### Provide advice and facilitate assistance

15. In accordance with paragraph 36 (a) of the annex to decision I/7, in consultation with the Party concerned, the Committee may provide advice and facilitate assistance to individual Parties regarding the implementation of the Convention. Subject to the Party's agreement, the Committee may also make recommendations to the Party concerned.<sup>7</sup>

#### Report to the Meeting of the Parties

16. The Committee submits a report on its activities to each ordinary session of the Meeting of the Parties and makes such recommendations as it considers appropriate.<sup>8</sup> The report provides an overview of the submissions, referrals, communications and requests considered by the Committee during the intersessional period as well as on the

<sup>&</sup>lt;sup>6</sup> Ibid., para. 14.

<sup>&</sup>lt;sup>7</sup> Ibid., para. 36 (b).

<sup>&</sup>lt;sup>8</sup> Ibid., para. 35.

implementation of decisions on compliance by individual Parties adopted by the Meeting of the Parties at its previous session. The report also highlights general compliance issues identified by the Committee in its work as well as commenting on the Parties' implementation of the Convention's reporting requirements. The report annexes a table of provisions of the Convention alleged or found not to have been complied with during the intersessional period.

17. During its preparation, the Committee's draft report is discussed in open session and posted on the Committee's website and Parties and observers are invited to provide their input. The Committee finalizes the report at least twelve weeks before the session of the Meeting of the Parties at which it is to be considered.<sup>9</sup>

#### **Monitor reporting requirements**

- 18. In accordance with paragraph 13 of the annex to decision I/7, the Committee's functions include to monitor, assess and facilitate the implementation of and compliance with the reporting requirements under article 10(2) of the Convention.
- 19. Article 10(2) of the Convention requires the Meeting of the Parties to keep under continuous review the implementation of the Convention on the basis of regular reporting by the Parties. In accordance with decision I/8 and IV/4 on reporting requirements, each Party is expected to report on the legislative, regulatory or other measures it has taken to implement the provisions of the Convention as well as the practical implementation of those measures. National implementation reports should be prepared through a transparent and consultative process involving the public. They should be submitted to the secretariat no later than 120 days before the session of the Meeting of the Parties for which they are submitted. In advance of each subsequent session of the Meeting of the Parties, each Party must review its report and submit an updated version of it to the secretariat. The secretariat is then tasked with preparing a synthesis report for each session of the Meeting of the Parties, summarizing the progress made and identifying significant trends, challenges and solutions. As part of its role to monitor the reporting requirements, the Committee reviews and provides comments on the draft synthesis report prepared by the secretariat.

## Measures by the Meeting of the Parties to address compliance

- 20. Upon consideration of the Committee's report and recommendations concerning a Party's non-compliance, the Meeting of the Parties decides on appropriate measures to bring about full compliance with the Convention. Depending on the particular question before it and taking into account the cause, degree and frequency of non-compliance, the Meeting of the Parties may decide to take one or more of the following measures:
  - (a) Provide advice and facilitate assistance to individual Parties regarding the

9

<sup>&</sup>lt;sup>9</sup> Ibid., para. 35.

- implementation of the Convention;
- (b) Make recommendations to the Party concerned;
- (c) Request the Party concerned to submit a strategy, including a time schedule, to the Compliance Committee regarding the achievement of compliance with the Convention and to report on the implementation of this strategy;
- (d) In cases of communications from the public, make recommendations to the Party concerned on specific measures to address the matter raised by the member of the public;
- (e) Issue declarations of non-compliance;
- (f) Issue cautions;
- (g) Suspend, in accordance with the applicable rules of international law concerning the suspension of the operation of a treaty, the special rights and privileges accorded to the Party concerned under the Convention;
- (h) Take other non-confrontational, non-judicial and consultative measures as may be appropriate.<sup>10</sup>

## Inter-sessional measures by the Committee to address compliance

21. Pending consideration by the Meeting of the Parties and with a view to addressing compliance issues without delay, the Compliance Committee may, in consultation with the Party concerned, provide advice and facilitate assistance regarding the implementation of the Convention (see measure (a) in para. 20 above). Also, the Committee may with the agreement of the Party concerned take the measures (b)-(d) in paragraph 20 above. In practice, most Parties who are subject to an examination by the Committee agree to receive recommendations directly from the Committee in this way, should they be found non-compliant.

<sup>&</sup>lt;sup>10</sup> Ibid., para. 37.

<sup>&</sup>lt;sup>11</sup> Ibid., para. 36(a).

<sup>&</sup>lt;sup>12</sup> Ibid., para. 36(b).

#### III. Committee's working methods

22. In addition to the rules on the Compliance Committee's working method set out in decision I/7, the Committee applies the rules of procedure of the Meeting of the Parties<sup>13</sup> mutatis mutandis. Rules 19, 20, 24 to 27, 29 to 42, 44, 46 and 48 of those rules are considered to be the most relevant to the Committee.<sup>14</sup>

#### Access to information about cases before the Committee

- 23. In keeping with the Convention's focus on transparency, except as explained below, all documents and correspondence, including email messages, received by or issued by the Committee or secretariat with respect to the substance of pending cases are posted on the Committee's website, www.unece.org/env/pp/cc.html, under the relevant case reference.
- 24. For reasons of personal privacy, to the extent feasible, home addresses, telephone numbers and emails of private persons will be redacted from documents and correspondence received or issued by the Committee prior to posting on the website. Parties, communicants and observers are accordingly requested to redact such details from their documentation prior to submitting them to the Committee.
- 25. If a Party, communicant or observer submits information to the secretariat requesting that information be considered by the Committee in connection with a particular case, but the Chair decides in consultation with the secretariat that the information submitted has no relevance to the compliance matters raised in the case, the information will not be circulated to the Committee, parties or posted on the website.
- 26. In addition, offensive or derisory language will not be posted on the website. Where the presence of such language comes to the attention of the Committee, it will be redacted and the rest of the document made available on the website.
- 27. If a communicant, Party or member of the public considers that a document submitted to the Committee contains information that should be kept confidential in accordance with the exceptions from disclosure set out in article 4(3) and (4) of the Convention, they may request the Committee to redact the relevant information from the version of the document published on the website. Pending a determination by the Chair on the request, the secretariat will promptly redact the information from the version of the document on the website and invite the other party to comment on the requested non-disclosure. If, after taking into account the request and the comments thereon, the Chair considers that the character of the information would indeed fall within the scope of an exception to disclosure in article 4(3) and (4) of the Convention, it will remain redacted

<sup>&</sup>lt;sup>13</sup> See the annex to decision I/1 of the Meeting of the Parties (ECE/MP.PP/2/Add.2).

<sup>&</sup>lt;sup>14</sup> MP.PP/C.1/2003/2, para. 11.

from the version of the document made available on the website. Alternatively, if the Committee considers that the relevant information would not be of a character to come within the scope of those exceptions, the document will be re-published in full.

28. Finally, if a communicant, Party, member of the public or the secretariat considers that any person may be penalised, persecuted or harassed due to any reference identifying that person in any documents before the Committee, they may request the Committee to redact all references that would identify that person. If the Chair decides that such redactions should be made, the references to that person will be deleted entirely from the documentation, including the versions circulated to the Committee and the parties as well as those posted on the website (see further paras. 245-249 below).

#### Open sessions with public participation

- 29. Meetings of the Committee are open to Parties, communicants, and members of the public as observers, except for the sessions of the meetings which, in accordance with paragraphs 26, 27, 29, 30 and 33 of the annex to decision I/7, should be closed.<sup>15</sup>
- 30. Specifically, hearings and discussions on particular cases are in general open to the public, and so are the open sessions concerning preliminary admissibility of new communications. Sessions at which the Committee discusses its working methods are also open to the public. Parties, communicants and observers may participate at all of these sessions. Participation is broadly understood in the sense enshrined in the Convention, comprising the right to comment and the right to have comments taken into account by the Committee, within the framework of the procedures of the meeting. 17
- 31. In addition to open sessions at its regular meetings, the Committee also from time-to-time holds open dialogue sessions, for example, at sessions of the Meeting of the Parties, during which Parties, NGOs and other members of the public are invited to make proposals on how the Committee's working methods may be improved.

#### NGOs and the Committee<sup>18</sup>

- 32. The fundamental role played by NGOs in the drafting of the Convention, as well as the role they now play in implementing it, is acknowledged, inter alia, in article 2(4) and (5) of the Convention. NGOs can submit communications to the Committee for its consideration under paragraph 18 of the annex to decision I/7 like any other member of the public. Moreover, NGOs falling with the scope of article 10(5) of the Convention and promoting environmental protection may nominate candidates for election to the Committee.<sup>19</sup>
- 33. The Committee welcomes the engagement of the NGO community. Beyond the

<sup>&</sup>lt;sup>15</sup> MP.PP/C.1/2003/2, para. 15.

<sup>&</sup>lt;sup>16</sup> Annex to decision I/7, para. 32.

<sup>&</sup>lt;sup>17</sup> MP.PP/C.1/2003/2, para. 16.

<sup>&</sup>lt;sup>18</sup> MP.PP/C.1/2003/4, paras. 27-33.

<sup>&</sup>lt;sup>19</sup> Annex to decision I/7, para. 4.

submission of communications on individual cases, NGOs can significantly contribute to the Committee's efforts to collect information by submitting information as observers. Paragraph 25 of the annex to decision I/7 does not make any distinction between information received from individuals, NGOs or Parties. NGOs may also contribute as "experts and advisers" whose services the Committee may seek. Most importantly, NGOs can make an important contribution to the follow-up on Committee findings and decisions and requests of the Meeting of the Parties on compliance, in particular by providing relevant and timely information on the extent to which the recommendations set out in the findings, decision or request have been implemented by the Party concerned.

## Use of audio and video conferencing to facilitate participation

- 34. With the dual aims of making its processes more participatory and reducing costs, the Committee uses audio conferencing, and where feasible, video and web conferencing to facilitate the participation of Parties, communicants and observers in its meetings.
- 35. The Committee offers Parties, communicants and observers the possibility to participate by audio conference in its open sessions on preliminary admissibility of communications and the follow-up on the Committee's findings and decisions and requests of the Meeting of the Parties on compliance.
- 36. The Committee has made clear, however, that audio conferencing, web conferencing and videoconferencing are not to be used for the hearing of a communication, submission or request from the Meeting of the Parties, and representatives of the Party concerned, submitting Party and communicants are therefore expected to attend the hearing of each communication, submission or request in person.<sup>20</sup>

#### **Working language**

37. English is the internal working language of the Committee.<sup>21</sup>

#### **Translation**

38. In order to be considered by the Committee, all documentation must be submitted in one of the Convention's official languages, i.e. English, French or Russian. As English is the working language of the Committee, if documentation is not submitted in English, this may considerably delay the process of its consideration as the secretariat will need to arrange for its translation before it can be considered by the Committee. Thus, to

<sup>&</sup>lt;sup>20</sup> MP.PP/C.1/2015/9, para. 80, ECE/MP.PP/2017/31, para. 29.

<sup>&</sup>lt;sup>21</sup> MP.PP/C.1/2003/2, para. 21.

ensure timely consideration by the Committee, submission of all documentation in English is encouraged.

- 39. The Committee will not examine documents submitted in languages other than English, French or Russian. Accordingly, in order for the Committee to consider documents in other languages, the Party, communicant or observer submitting the documentation should provide a translation, preferably in English and submit it together with the document in its original language. For transparency, and in particular so that interested members of the public in the Party concerned can follow the procedure, the document should be submitted in its original language also. Certified translations are preferable to unofficial. In some situations the Party concerned may be asked by the Committee to provide the translation, particularly of legislation and caselaw.
- 40. For supporting documentation, for example, legislation and caselaw, it may not be necessary to translate the entire document. However, a translation of any pertinent sections of the document that should be considered by the Committee must be provided together with the document. The Party, communicant or observer submitting the document should also clearly explain in its accompanying submissions the relevance of the document to the case.

#### **Use of hyperlinks**

41. All documents, including supporting documents, should be provided in pdf format. The Committee will not take into account information provided only through hyperlinks. This is for reasons of transparency, fairness and due process.

#### **Publication of meetings and documentation**

- 42. Meetings of the Committee are publicized on the Convention's website, <sup>22</sup> with the provisional meeting agenda, provisional meeting timeline, meeting reports, official documents and any draft documents proposed for discussion in open session also being posted there. <sup>23</sup>
- 43. As far as possible, informal discussion papers prepared by the Chair or secretariat for a meeting of the Committee will be posted on the webpage for that meeting approximately one week in advance of the meeting.
- 44. All documentation submitted to or issued by the Committee or secretariat, and all documentation submitted by the Parties concerned, submitting Parties, communicants and observers are made available on the website, other than any information which is required to be kept confidential pursuant to Chapter VIII of the annex to decision I/7 (see paras. 245-249 below).
- 45. Once the Chair and Vice-Chair(s) of the Committee have decided that a

<sup>&</sup>lt;sup>22</sup> http://www.unece.org/env/pp/ccmeetings.html

<sup>&</sup>lt;sup>23</sup> MP.PP/C.1/2003/2, para. 18.

communication should be forwarded to the Committee for consideration as to its preliminary admissibility, the communication, together with any supporting documentation, is published on the website and the Party concerned is informed (see paras. 88-96 below).

- 46. Publication on the website is intended to facilitate the public to access information relating to Parties' compliance with the Convention and does not imply endorsement of the content by the Committee or by the UNECE. Nor does the posting of any document provided by a party or observer to a case in any way prejudice the Committee's findings or indicate any particular view of the Committee.
- 47. Draft findings and recommendations agreed by the Committee are posted on the website once they have been transmitted to the Party concerned and the submitting Party or the communicant (i.e. the parties concerned). Similarly, any comments provided by the parties concerned or observers will be made publicly available on the website.<sup>24</sup>

#### Closed sessions

48. The Committee's deliberations during the preparation and adoption of any findings (including determinations on preliminary admissibility, and findings regarding the implementation of its recommendations as well as of requests and decisions of the Meeting of the Parties on compliance), measures or recommendations are held in closed session.<sup>25</sup>

#### **Virtual meetings**

49. In order to ensure the efficient management of its workload, the Committee regularly holds virtual meetings between its ordinary meetings. During virtual meetings, the Committee may discuss any aspect of its work that would be discussed in closed session at an ordinary meeting. This includes the preparation, finalization and adoption of findings and recommendations and its reviews of the implementation of its recommendations as well as of requests and decisions of the Meeting of the Parties on compliance.

#### **Decision-making**

50. The participation of five members of the Committee is required for any decision to be taken. The Committee makes every effort to reach its decisions by consensus. Decisions of a procedural nature can be taken by a simple majority of the members present and voting. Decisions on substantive matters can be taken only with the support of seven out of nine members present and voting; six out of eight members present and voting; six out of seven members present and voting; five out of six members present and voting; and four out of five members present and voting. Notwithstanding this, the

<sup>&</sup>lt;sup>24</sup> MP.PP/C.1/2004/8, para. 35, and ECE/MP.PP/C.1/2005/8, para. 25.

<sup>&</sup>lt;sup>25</sup> MP.PP/C.1/2003/2, para. 17, and the annex to decision I/7, para. 33.

Committee is generally sympathetic to the view that at least five members should be in support of any substantive decision being taken. Since Committee members are elected strictly in their personal capacity, neither the Committee nor the absent Committee member may designate a substitute.<sup>26</sup>

#### **Decision-making by email**

- 51. In order to ensure the efficient management of its workload, the Committee may take decisions between its meetings by email. Generally, the Committee's electronic decision-making procedure is used for decisions that do not require extensive further discussion by the Committee. Depending on the particular circumstances, such decisions may include, inter alia, the preparation and completion of draft findings; the finalization and adoption of findings, the adoption of progress reviews and meeting reports, determinations on the admissibility of communications, and the drafting of questions to be raised with the parties to a case.
- 52. The Committee's electronic decision-making procedure is as follows:
  - (a) The Chair prepares, with the assistance of the secretariat and, depending on the nature of the decision, the curator (see paras. 66 and 67 below), the draft text to be decided through electronic decision-making. The Chair decides on a suitable time period for the Committee members to respond. The secretariat circulates the draft text to the Committee members by email, specifying the deadline for response.
  - (b) Committee members are asked to reply by a given date whether he or she is satisfied with the text proposed, or to propose amendments.
  - (c) Once all Committee members have indicated their satisfaction with the Chair's (latest) proposal, or alternatively, if no member of the Committee objects to it by a specified deadline, the proposal is deemed adopted by the Committee.
  - (d) If based on the comments received it becomes clear that the matter is not suitable for electronic decision-making, the Chair may decide to postpone the decision-making procedure to the next ordinary or virtual meeting.
- 53. At the instigation of the Chair, the Committee may also use other forms of communication, such as conference calls, possibly in combination with email, to take decisions to ensure the Committee's efficient functioning between meetings.

#### **Missions**

54. To assist the performance of its functions and subject to availability of resources, the Committee may from time to time undertake a mission to the territory of a Party or

<sup>&</sup>lt;sup>26</sup> MP.PP/C.1/2003/2, para.12; ECE/MP.PP/C.1/2008/6, para. 42.

another State. Such missions will only be undertaken with the agreement of that Party or State. Missions may be undertaken to gather information to assist the Committee in its examination of compliance (see further paras. 164-177 below).<sup>27</sup> They may also be undertaken after the adoption of findings and recommendations, in order to provide advice and facilitate assistance to individual Parties regarding the implementation of the Convention.<sup>28</sup> Finally, the Committee may undertake missions at the invitation of a Party to provide information on the Convention and the Committee's activities.

#### Role of the secretariat

55. According to the annex to decision I/7, the secretariat shall arrange for and service the meetings of the Committee.<sup>29</sup> The secretariat has the task of ensuring that meetings are well prepared and documented. This includes supporting the Committee with the preparation of relevant documents before and at the meeting as well as ensuring the proper documentation of the meeting's outcomes. The secretariat facilitates the smooth management of the Committee's caseload between meetings by preparing outgoing correspondence in accordance with the Committee's instructions, processing incoming correspondence from parties and observers, and liaising with Parties, communicants and observers. Finally, the secretariat provides guidance upon request to Parties, communicants and members of the public regarding the Committee's procedures.

<sup>&</sup>lt;sup>27</sup> Annex to decision I/7, para. 25(b).

<sup>&</sup>lt;sup>28</sup> Ibid., paras. 36(a) and 37(a).

<sup>&</sup>lt;sup>29</sup> Ibid., para. 12.

#### IV. Composition and election

#### **Composition of the Committee**

56. The Compliance Committee consists of nine individuals<sup>30</sup> serving in their personal capacity.<sup>31</sup> The Committee shall be composed of nationals of the Parties and Signatories to the Convention who shall be persons of high moral character and recognized competence in the fields to which the Convention relates, including persons with legal experience.<sup>32</sup>

#### **Election of Committee members**

#### **Nomination**

- 57. Unless the Meeting of the Parties, in a particular instance, decides otherwise, the procedure for the nomination of candidates for the Committee is as follows:<sup>33</sup>
  - (a) Candidates meeting the requirements described in paragraph 56 above may be nominated by Parties, Signatories and non-governmental organizations falling within the scope of article 10(5) of the Convention and promoting environmental protection;
  - (b) Nominations are to be sent to the secretariat in at least one of the official languages of the Convention at least twelve weeks before the opening of the session of the Meeting of the Parties at which the election is to take place;
  - (c) Each nomination must be accompanied by a curriculum vitae (CV) of the candidate not exceeding 600 words and may include supporting material;
  - (d) The secretariat then distributes the nominations, including the CVs, together with any supporting material.

#### **Election**

58. The Meeting of the Parties elects the members of the Committee by consensus or, failing consensus, by secret ballot.<sup>34</sup>

<sup>30</sup> Decision II/5, para. 12.

<sup>&</sup>lt;sup>31</sup> Annex to decision I/7, para. 1.

<sup>&</sup>lt;sup>32</sup> Ibid., para. 2.

<sup>33</sup> Ibid., para. 5.

<sup>&</sup>lt;sup>34</sup> Ibid., para. 7.

#### **Geographic representation**

59. The Committee may not include more than one national of the same State.<sup>35</sup> In the election of the Committee, consideration is given to the geographical distribution of membership and diversity of experience.<sup>36</sup>

#### **Rotation**

- 60. A full term of office starts at the end of the ordinary session of the Meeting of the Parties at which the member is elected (or re-elected) and runs until the close of the second ordinary session of the Meeting of the Parties thereafter.<sup>37</sup>
- 61. At its first session, the Meeting of the Parties elected four members to the Committee to serve until the end of its next ordinary session and four members to serve a full term of office.<sup>38</sup> At its second session, the Meeting of the Parties agreed that the number of members of the Committee should be increased from eight to nine; this change took effect at the third ordinary session of the Meeting of the Parties, at which five members were elected for a full term. At each ordinary session thereafter, the Meeting of the Parties elects four or five members, as appropriate, for a full term of office.
- 62. Outgoing members may be re-elected once for a further full term of office, unless the Meeting of the Parties decides otherwise. The Committee elects its own Chair and Vice-Chair(s).<sup>39</sup>
- 63. If a member of the Committee for some reason can no longer perform his or her duties, the Bureau of the Meeting of the Parties appoints another member who fulfils the criteria described above to serve the rest of the term, subject to the approval of the Committee.<sup>40</sup>

<sup>&</sup>lt;sup>35</sup> Ibid., para. 3.

<sup>&</sup>lt;sup>36</sup> Ibid., para. 8.

<sup>&</sup>lt;sup>37</sup> Ibid., para. 9.

<sup>&</sup>lt;sup>38</sup> Ibid.

<sup>&</sup>lt;sup>39</sup> Ibid.

<sup>&</sup>lt;sup>40</sup> Ibid., para. 10.

## V. Responsibilities of Committee members

#### Impartiality and conscientiousness

64. Every member serving on the Committee must, before taking up his or her duties, make a solemn declaration in a meeting of the Committee that he or she will perform his or her functions impartially and conscientiously.<sup>41</sup> For the sake of clarity, the Committee requires incoming members to sign this declaration in writing.

#### **Collective decision-making**

65. The Committee takes its decisions collectively and the Committee members are thus collectively responsible for the Committee's decisions. This means that all Committee members are expected to read all relevant information for all cases.

#### The role of the curator

- 66. To facilitate the efficient functioning of the Committee, the Chair assigns each case to a Committee member who thereafter acts as the curator for the case. All cases including communications, submissions, referrals, requests from the Meeting of the Parties or requests from a Party for advice or assistance are assigned in this way. A curator is also assigned to each decision and request of the Meeting of the Parties concerning compliance by individual Parties.
- 67. The responsibilities of the curator include drafting questions to be sent by the Committee to the parties in a case, taking the lead in proposing questions to the parties during the hearing of the case, preparing the draft findings and recommendations for the Committee's deliberations and, in cases where non-compliance is found, reviewing the implementation of the Committee's findings and recommendations by the Party concerned. While the curator's work ensures the smooth progress of the case, any decision is to be taken by the Committee collectively.

#### **Conflict of interest**

68. In a case where a Committee member finds himself or herself faced with a possible or apparent conflict of interest, that member is required to bring the issue to the Committee's

20

<sup>&</sup>lt;sup>41</sup> Ibid., para. 11.

attention before the Committee's consideration of the case at issue.

- 69. Specifically, the Committee has agreed on the following guiding principles on the independence and impartiality of Committee members:<sup>42</sup>
  - (a) The Committee members shall exercise their functions independently and impartially, free from interference or influence by any source and from the appearance of interference or influence;
  - (b) The Committee shall decide cases impartially, on the basis of the facts of the case and the applicable law;
  - (c) In relation to pending cases, including cases following up on requests and decisions of the Meeting of the Parties on compliance, Committee members must avoid situations that might give rise to a conflict of interest or which might reasonably be perceived by the Parties or by members of the public as giving rise to such a conflict;
  - (d) Being a citizen of the Party whose compliance is to be discussed will not in itself be considered as a conflict of interest.<sup>43</sup>
  - (e) A Committee member shall disclose to the Committee, at the next meeting, or sooner if appropriate, any circumstances which could reasonably be considered as leading to a conflict of interest or which might reasonably be perceived by the Parties or by members of the public as giving rise to such a conflict. Such circumstances may include a Committee member's relationship with the Party concerned, with the communicant, or with an observer who has made submissions in the case. It may also include a Committee member's past dealings with the case itself. Before bringing the issue to the attention of the Committee, the Committee member may consult with the Chair.
  - (f) Upon the disclosure of a conflict of interest, the Committee will consider the information provided and take appropriate action. If in doubt as to whether or not a situation might give rise to an actual conflict or be perceived by a reasonable Party or member of the public as a conflict, Committee members should err on the side of caution.
- 70. Any member considered to have a possible conflict of interest will be able to attend sessions and access information in the same manner as an observer related to that particular matter. While the Committee member can thus attend the open sessions and discussions, he or she is expected not to actively participate in the discussion. A Committee member who has a conflict of interest may not take part in closed sessions where the Committee is considering compliance or preparing or adopting findings, measures or recommendations.<sup>44</sup>

<sup>&</sup>lt;sup>42</sup> ECE/MP.PP/C.1/2010/6, para. 6.

<sup>&</sup>lt;sup>43</sup> MP.PP/C.1/2003/2, para. 22.

<sup>&</sup>lt;sup>44</sup> MP.PP/C.1/2004/6, para. 53.

- 71. The rules on conflict of interest also apply to any electronic decision-making procedure.
- 72. Committee members are not excluded from providing advice in response to queries from non-governmental organizations (NGOs) or others who consider submitting a communication to the Committee. However, it is recommended that the Committee member refer the person/NGO to the information available on the website or to the secretariat. This is to avoid Committee members specifically advising such individuals and/or organizations, which could, in some cases, lead to a conflict of interest for that member of the Committee later on. For this reason, any compliance-related correspondence or queries should be sent to the secretariat and not to individual members of the Committee.<sup>45</sup>
- 73. Subject to the provisions of the annex to decision I/7 related to confidentiality, Committee members are at liberty to deal with requests for information about submissions, referrals and communications under consideration where such information is already in the public domain. However, channelling such requests via the secretariat will ensure more up-to-date and complete information and is therefore recommended.<sup>46</sup>
- 74. Members of the Committee may accept invitations to present the compliance mechanism at appropriate events, including conferences and workshops,<sup>47</sup> or to participate in capacity-building activities and projects related to the Convention, e.g. as expert consultants. Unless specifically agreed by the Committee, Committee members presenting or discussing the Committee or its work in such contexts act in their individual capacity and do not represent the Committee. The participation of Committee members in their individual capacity in such events and activities does not in itself create a conflict of interest, but it is possible that in some cases such involvement may lead to a conflict of interest at a later stage; such a case may for instance be created if a Committee member has provided expert assistance in the development of relevant legislation and a case relating to this legislation is later subject to review before the Committee. Should such situations arise, the standard procedures apply and the Committee member concerned is expected to notify the Committee of any potential conflict of interest.<sup>48</sup>

<sup>&</sup>lt;sup>45</sup> MP.PP/C.1/2003/4, para 8.

<sup>&</sup>lt;sup>46</sup> MP.PP/C.1/2004/6, para 49.

<sup>&</sup>lt;sup>47</sup> MP.PP/C.1/2003/4, para. 30.

<sup>&</sup>lt;sup>48</sup> ECE/MP.PP/C.1/2007/4, para. 28.

# VI. Overview of compliance review – step by step

75. The purpose of this section of the Guide is to provide an overview of the stages in the Committee's consideration of a case.

#### **Triggering review of compliance**

76. The various ways in which the compliance mechanism can be triggered are explained in paragraph 83 below.

#### **Determination of preliminary admissibility**

77. In the case of a communication, before its substance is considered by the Committee, it must first be determined to be preliminary admissible. This is explained in paragraphs 86-120 below.

#### Response by the Party concerned

78. The requirements for the response by the Party concerned to a communication or submission concerning its compliance are explained in paragraphs 125-133 and 142-147 below.

#### Information gathering

79. Once the response has been received from the Party concerned, and in some cases beforehand, the Committee may determine that it needs to gather further information regarding the matters at issue in the case. The Committee's powers of gathering information are explained in paragraphs 164-177 below.

#### Where needed, a hearing with the parties concerned

80. Under paragraph 24 of the annex to decision I/7, the Committee may hold hearings. It also has the discretion to complete its deliberations on a case without holding a hearing. The Committee's procedure regarding hearings is explained in paragraphs 182-194 below.

#### **Preparation of findings**

81. Once the Committee has gathered the information it considers needed, it will commence the preparation of its findings. The preparation of findings is discussed in paragraphs 195-203 below.

#### Follow-up if non-compliance found

82. The procedure with respect to follow-up on findings of non-compliance is described in paragraphs 204-223 below.

#### VII. Triggering review of compliance

#### **Types of compliance cases**

- 83. As described in paragraph 4 above, the Committee's review of a Party's compliance may be triggered in five ways:
  - (a) Members of the public may make communications concerning a Party's compliance with the Convention;
  - (b) A Party may make a submission about compliance by another Party;
  - (c) A Party may make a submission concerning its own compliance;
  - (d) The secretariat may make a referral to the Committee;
  - (e) The Meeting of the Parties may request the Committee to examine a Party's compliance with the Convention.<sup>49</sup>
- 84. Thus far the vast majority of the Committee's caseload has been triggered by communications from the public. This section first describes how such communications are processed, and then describes the process for Party submissions, referrals by the secretariat, and lastly, requests by the Meeting of the Parties regarding a Party's compliance.
- 85. Those elements of the Committee's procedure for reviewing compliance which apply to all the types of cases in its caseload are discussed in paragraphs 164-223 below.

25

<sup>&</sup>lt;sup>49</sup> This trigger is not expressly referred to in the annex to decision I/7, but it follows from the Meeting of the Parties' general decision-making capacity.

#### (a) Communications from the public

86. Communications may be brought before the Committee by one or more members of the public concerning a Party's compliance with the Convention. The communication shall be addressed to the secretariat in writing using the required format,<sup>50</sup> and should be submitted in electronic form supported by corroborating information.

#### Upon receipt of a new communication

87. Following receipt of a new communication, the secretariat sends an acknowledgement of receipt. The secretariat checks it for completeness and determines whether it is sufficiently in the form of a communication to be brought to the attention of the Chair and Vice-Chair(s) of the Committee. If not, the secretariat informs the communicant accordingly and invites the communicant to re-submit its communication in the required format.<sup>51</sup>

### Decision by Chair and Vice-Chair(s) to forward communication to the Committee for review

- 88. Approximately six weeks before each Committee meeting, the secretariat forwards all communications received since the last meeting that are sufficiently in the form of a communication to the Chair and Vice-Chair(s) of the Committee for their review.
- 89. The Chair and the Vice-Chair(s), with the assistance of the secretariat, decide whether the forwarded communications are sufficiently well-prepared to be considered by the Committee at its next meeting for a preliminary determination on admissibility. If not, the secretariat informs the communicant accordingly and invites the communicant to re-submit the communication in the required format for communications.
- 90. The decision by the Chair and Vice-Chair(s) that a new communication should be forwarded to the Committee for consideration of preliminary admissibility does not in any way prejudice the Committee's finding on whether the communication in question is preliminarily admissible. Moreover, the decision to forward a communication to the Committee does not prevent the Committee from deciding to request further clarification by the communicant before deciding on the question on preliminary admissibility.
- 91. Each communication forwarded to the Committee for consideration of preliminary admissibility will be allocated a case reference, in the format "PRE/ACCC/C/[Year]/[Number of communication]". The prefix "PRE" indicates that the Committee's determination of the preliminary admissibility of that communication has not yet been made.

<sup>&</sup>lt;sup>50</sup> See annex I to this Guide. The required format is available in electronic format on the Committee's website at http://www.unece.org/env/pp/cc/com.html <sup>51</sup> Ibid.

92. The Committee agrees at each meeting the cut-off date, at least six weeks before the next Committee meeting, by which the Committee will receive communications to be considered for the next meeting. This date will be recorded in the meeting report and then posted on the communications webpage. Only communications submitted to the secretariat by this date will be considered by the Chair and Vice-Chair(s) for the next Committee meeting. Communications received after the six-week cut-off date may be considered for preliminary admissibility at the following meeting.

#### **Determination of admissibility**

#### New communications published on the website and Party concerned notified

- 93. Not later than four weeks before each Committee meeting, the communications to be considered for preliminary admissibility by the Committee at that meeting will be posted, together with any attachments, on the webpage for communications, http://www.unece.org/env/pp/cc/com.html.
- 94. Also no later than four weeks before the meeting, the secretariat informs the Party concerned that a communication concerning its compliance will be considered as to its preliminary admissibility at the next meeting and provides a link to where the communication is posted on the Committee's website. The Party concerned is informed that the preliminary admissibility will be discussed in open session at the upcoming meeting and, though there is no requirement that it do so, a representative of the Party concerned may participate either by audio-conference or in person in that session.
- 95. At the same time, the secretariat informs the communicants whose communications will be considered as to their preliminary admissibility at the upcoming meeting of that fact and refers to the information to be posted on the Committee's website. The communicant is informed that the preliminary admissibility will be discussed in open session at the upcoming meeting and, though there is no requirement that it do so, the communicant may participate either by audio-conference or in person in that session.
- 96. Also at this time, the secretariat informs the Committee that the communications to be considered for preliminary admissibility at its next meeting have been posted.

#### Criteria for determining preliminary admissibility

- 97. Paragraph 20 of the annex to decision I/7 states that the Committee shall consider any such communication unless it determines that the communication is:
  - (a) Anonymous;
  - (b) An abuse of the right to make such communications;
  - (c) Manifestly unreasonable;
  - (d) Incompatible with the provisions of decision I/7 or with the Convention.

- 98. In addition, paragraph 19 of the annex to decision I/7 requires the communication to be supported by corroborating information.<sup>52</sup>
- 99. According to paragraph 21 of the annex to decision I/7, the Committee should also at all relevant stages take into account any available domestic remedy unless the application of the remedy is unreasonably prolonged or obviously does not provide an effective and sufficient means of redress.<sup>53</sup> In this regard, the very high cost of the domestic procedures may be relevant when assessing whether a domestic remedy is in fact available in practice.
- 100. In the light of the purpose and the functions of the Committee to promote and improve compliance by the Parties with the Convention, the Committee may determine that a communication, while broadly appearing to fulfil the admissibility requirements of paragraph 20 of the annex to decision I/7, after careful consideration does not pass a threshold of *de minimis*. In such cases, the Committee may find the communication inadmissible under paragraph 20(d) of the annex to decision I/7 on the ground that by not passing a threshold of *de minimis* with respect to communication's relevance and importance in the light of the purpose and functions of the Committee, the communication is incompatible with the provisions of decision I/7. This criterion has developed in the Committee's practice and it has been welcomed by the Meeting of the Parties.<sup>54</sup>

#### Procedure for determining preliminary admissibility

- 101. The Committee's procedure for determining preliminary admissibility during a meeting follows the sequence of the Committee's procedure for discussing the substance of a communication. This means that the Committee first discusses the communication in open session, and then deliberates in closed session.
- 102. During the open session, after inviting the curator of the communication to provide a brief introduction to the communication, the Chair gives the floor to any representatives of the Party concerned and communicant present either in person or via audio-conference that wish to briefly state their view regarding the admissibility of the communication. This is without prejudice to the right of the Party concerned to make further submissions on the admissibility of the communication at any point prior to the Committee's final determination on admissibility (see below). The lack of a statement by the Party concerned at the meeting to discuss preliminary admissibility does not preclude it making submissions on admissibility later.
- 103. Observers may also be given an opportunity to briefly state their views on the admissibility of the communication.
- 104. At this stage, all statements should be concise and strictly limited to the issue of admissibility, leaving aside the substance of the communication. Statements should be no

<sup>&</sup>lt;sup>52</sup> Annex to decision I/7, para. 19.

<sup>&</sup>lt;sup>53</sup> Ibid., para. 21.

<sup>&</sup>lt;sup>54</sup> ECE/MP.PP/2011/11, paras. 18 and 19, and decision IV/9, para. 2.

more than 2-3 minutes in length. All parties are encouraged to provide their statements in writing to the secretariat at least one week in advance of the meeting.

- 105. Following the short statements, the Committee may put questions to representatives of the Party concerned or communicant or to observers participating in the session.
- 106. The Committee then deliberates upon each communication in closed session. During the closed session, it may proceed to prepare its preliminary determination of admissibility or adopt a finding that the communication is not admissible. Alternately, it may agree to defer its determination of preliminary admissibility in order to request the communicant to clarify, further substantiate or re-structure its communication prior to the next meeting or to seek clarification from the Party concerned, e.g. regarding the availability of further domestic remedies.
- 107. Once a communication is determined to be preliminarily admissible or found to be inadmissible, the prefix "PRE" is removed from its case reference.
- 108. The Committee's determination of preliminary admissibility, its finding that the communication is inadmissible or its decision to defer its determination to request further clarification or information is announced in open session during the meeting usually in its last session and recorded in the meeting report.
- 109. If the Committee finds the communication to be inadmissible under paragraph 20 of the annex to decision I/7, the reasoning on which the Committee's finding is based is recorded in the meeting report.

#### Party concerned and communicant notified of outcome

- 110. No later than two weeks after the meeting, the secretariat informs the Party concerned and communicant about the Committee's decision regarding the preliminary admissibility of the communication. If the communication was found to be inadmissible, the communicant and the Party concerned will also be informed of the reason why.
- 111. If the communication was determined to be preliminarily admissible, the communication is formally forwarded by the secretariat to the Party concerned at the time of forwarding the Committee's determination of preliminary admissibility. The Party concerned has five months from the date that the secretariat forwards the Committee's determination of preliminary admissibility to provide its response (see paras. 125-127 below).
- 112. If the communication is found inadmissible, the case is considered to be closed. The communication and all related correspondence will remain available on the website.

#### Timeframe for final determination of admissibility

113. The Committee may reconsider its determination that the communication is preliminary

admissible at any stage during its examination up until the time of the hearing or, if the Committee has decided that a hearing is not needed, the time that the Committee informs the parties that it will commence its deliberations on the substance of the case.

#### Request for reconsideration of determination of inadmissibility

114. If the communicant considers that the Committee's determination that the communication is inadmissible was based on a manifest error, the communicant may request a reconsideration of the Committee's determination of inadmissibility. Any such request must be submitted to the secretariat, with a reasoning of no more than 1200 words (2 pages A4 size), no later than five weeks after the communicant was informed of the Committee's determination of inadmissibility, together with the reasons for its determination. The communicant's request for reconsideration will be forwarded by the secretariat to the Chair and Vice-Chair(s) who will consider whether the communicant's request for reconsideration identifies a possible manifest error. If they consider that the request for reconsideration does not identify a possible manifest error, the communicant will be informed accordingly and the Committee's earlier determination of inadmissibility will stand. If, however, the Chair and Vice-Chair(s) consider that the request for reconsideration may indeed identify a possible manifest error, the communicant's request will be sent to the Party concerned which will thereafter have three weeks to comment on the request. The Committee will thereafter make its decision on the communicant's request, taking into account any comments received from the Party concerned, at its next ordinary or virtual meeting. The Committee's decision on the communicant's request will be final and not subject to review or appeal.

#### Following a finding of inadmissibility

115. Subject to the procedure for reconsideration in paragraph 114 above, the Committee's finding that a communication is inadmissible is final and not subject to appeal or review.

#### Use of domestic remedies

116. The Committee will take into account, at all relevant stages, any available domestic remedy unless the application of the remedy is unreasonably prolonged or obviously does not provide an effective and sufficient means of redress.<sup>55</sup> It is at the discretion of the Committee to determine a communication to be inadmissible under paragraphs 20 and 21 of the annex to decision I/7 if in its view the communicant has not sufficiently used the available domestic administrative or judicial review procedures.

117. If no domestic remedies are available or the remaining domestic remedies would be unreasonably prolonged or would not provide an effective or sufficient means of redress, the

\_

<sup>&</sup>lt;sup>55</sup> Annex to decision I/7, para. 21.

communicant will need to provide appropriate evidence to the Committee to establish this fact. Possible evidence that a remedy is unreasonably prolonged or would not provide an effective and sufficient means of redress could be that the Committee has previously found that the relevant administrative and judicial procedures available in the Party concerned do not comply with article 9 of the Convention.

- 118. Since the compliance review mechanism is not a redress mechanism, if other members of the public have already exhausted the domestic remedies available to challenge the alleged non-compliance, then this may be taken into account by the Committee for the purposes of paragraph 21 of the annex to decision I/7. In this case, the communicant should provide the Committee with information on the final outcomes of the proceedings brought by those other persons to demonstrate that the available domestic remedies have indeed been exhausted.
- 119. Upon learning of the existence of a pending domestic review procedure, the Committee will ask the communicant to promptly provide it with clear reasons as to why, notwithstanding the pending domestic review procedure, the Committee should (depending on the stage of the communication) provisionally admit the communication or, if the communication was already determined preliminary admissible, uphold its earlier determination of preliminary admissibility. The Committee will thereafter consider any reasons provided by the communicant and any comments thereon from the Party concerned in the light of paragraphs 20 and 21 of the annex to decision I/7 and if it considers that the thresholds set out in those paragraphs have not been met, it may determine the communication to be inadmissible.
- 120. For clarity, domestic remedies are the remedies provided by the legal system of the Party concerned. If the communicant is from a country outside the Party concerned, it is still expected to use those domestic remedies that are available to it in the legal system of the Party concerned.

#### **Summary proceedings**

- 121. Bearing in mind that according to the Convention the compliance review mechanism is not a redress mechanism, and on the basis of paragraph 20 of the annex to decision I/7, in cases which have been determined to be preliminarily admissible, but where the legal issues raised by the communication have already been the subject of findings by the Committee, the Committee may decide to apply its summary proceedings procedure.
- 122. In such cases, the Committee will send a letter to the communicant and the Party concerned of the Committee's decision to apply its summary proceedings procedure.
- 123. A communicant whose communication is subject to summary proceedings will be included in any follow-up on the earlier case in which non-compliance on those issues was found, and will be given the opportunity to comment, together with the communicants of the earlier case.
- 124. The Committee's decision to apply its summary proceedings procedure will be recorded

in the meeting report and the Committee will also include in its report to the next session of the Meeting of the Parties that the communication was decided through its summary proceedings procedure.

#### Response by the Party concerned

- 125. When the Party concerned receives a letter from the secretariat forwarding a communication for its response, it should as soon as possible, but in any case no later than five months from the date of the secretariat's letter, submit written explanations or statements clarifying the matter and responding to the allegations.<sup>56</sup> In its response the Party concerned should explicitly comment on the allegations made in the communication and also address any questions and other points raised by the Committee at the time the communication was forwarded.
- 126. The Party concerned may also submit comments with respect to the admissibility of the communication. If a Party contests the admissibility of the communication, it should inform the Committee as soon as possible, but no later than five months from the date the communication was forwarded.
- 127. The five-month deadline for the response is calculated from the date the communication and any relevant documentation were forwarded to the Party concerned by the secretariat for its response. The response from the Party concerned should reach the secretariat by the end of the five-month period by email. The posted original may arrive after the lapse of the five-month period, as long as it was posted before the expiry of the deadline.
- 128. The response from the Party concerned should be no more than 6,000 words (ten A4 pages). If in an exceptionally complex case more than ten pages are required, in no circumstances should the response be longer than 12,000 words (twenty A4 pages). The paragraphs of the response should be numbered.
- 129. A list of annexes should be included at the end of the response. The response should clearly specify which paragraph of the response each annexed document relates to. The relevant parts of each annex should be highlighted.
- 130. The submission of extraneous, superfluous or bulky documentation should be avoided. Only documentation essential to the response should be submitted, including:
  - (a) Relevant national legislation, with the most relevant provisions highlighted.
  - (b) Relevant court decisions and the results of other review procedures, with the most relevant sections highlighted.
  - (c) Other relevant documentation which the Party concerned considers answers the allegations of non-compliance, with the most relevant sections highlighted.
- 131. All supporting documentation should be submitted in the original language, together with a legal standard English translation thereof, or if that is not possible, a legal standard

-

<sup>&</sup>lt;sup>56</sup> Annex to decision I/7, para. 23.

translation in either Russian or French.

- 132. Upon receipt of the response from the Party concerned, the Committee will consider any comments it has made with respect to the admissibility of the communication. If the Committee considers that these raise serious doubts concerning admissibility, the Committee will invite the communicant to comment on those submissions and may seek further information from either party. After taking into account the parties' comments, it may decide to uphold its preliminary decision of admissibility or reverse that decision and find the communication to be inadmissible. The parties will be informed of the outcome of the Committee's decision (see paras. 110-112 above). If the Party concerned does not challenge the admissibility of the communication, or the Committee does not find the Party concerned's arguments on this point to be persuasive, the Committee will consider whether any further information should be requested from either party prior to it commencing its deliberations on the substance of the communication.
- 133. If the Party concerned does not respond within the five-month deadline, the Committee will likewise proceed to consider the substance of the file, on the basis of the information before it.
- 134. If prior to commencing its deliberations on the substance of the communication the Committee decides that a hearing will be needed (see paras. 182-189 below), the Party concerned and the communicant will be informed of their right to be represented at the hearing in accordance with paragraph 32 of the annex to decision I/7. Subject to financial resources, financial support will be provided where needed to assist a representative of the communicant and an eligible government representative from the Party concerned to participate.

## (b) Submissions by Parties concerning other Parties

- 135. A submission may be brought before the Committee by one or more Parties that have reservations about another Party's compliance with its obligations under the Convention. Such a submission shall be addressed in writing to the secretariat and supported by corroborating information.
- 136. A submission by a Party concerning another Party's compliance should be no more than 6,000 words (ten A4 pages). If in an exceptionally complex case more than ten pages are required, in no circumstances should the submission be longer than 12,000 words (twenty A4 pages). The paragraphs of the submission should be numbered.
- 137. A list of annexes should be included at the end of the submission. The submission should clearly specify which paragraph of the submission each annexed document relates to. The relevant parts of each annex should be highlighted.
- 138. The submission of extraneous, superfluous or bulky documentation should be avoided. Only documentation essential to the submission should be submitted, including:
  - (a) Relevant national legislation, with the most relevant provisions highlighted.
  - (b) Relevant court decisions and the results of other review procedures, with the most relevant sections highlighted.
  - (c) Other relevant documentation which the submitting Party considers supports its allegations of non-compliance, with the most relevant sections highlighted.
- 139. All supporting documentation should be submitted in the original language, together with a legal standard English translation thereof, or if that is not possible, a legal standard translation in either Russian or French.

#### Upon receipt of a new submission

- 140. Within two weeks of receiving a submission, the secretariat shall send a copy of it to the Party whose compliance is at issue.<sup>57</sup>
- 141. When forwarding the submission, the secretariat will, in a cover letter, request the Party concerned to acknowledge receipt of the submission and remind it of its obligation under decision I/7 to reply within three months or such longer period as the circumstances of the particular case may require, but no later than six months from the date the submission was forwarded.

34

<sup>&</sup>lt;sup>57</sup> Annex to decision I/7, para. 15.

#### Response from the Party concerned

- 142. The deadline for the response of the Party concerned is calculated from the date at which the documentation was sent by the secretariat; and the response from the Party concerned should reach the secretariat by the end of the relevant period by email. The cover letter from the secretariat forwarding the submission invites the Party concerned to indicate whether, due to the circumstances of the particular case, the Party concerned envisages any difficulty in providing the reply within three months, and if so, to indicate when a reply would be sent. In the first instance, it is for the Party concerned to determine whether more than three months is necessary to provide a reply.
- 143. The response of the Party concerned should adhere to the format described in paragraphs 128-131 above.
- 144. If no substantive reply is received from the Party concerned after three months or such longer period as may have been specified by the Party concerned, the secretariat sends a reminder to the Party concerned. The reminder will point out that following the expiry of the six-month period, the Committee will be required to deal with the case on the basis of the information available to it, even in the absence of any response from the Party concerned. If necessary, a further and final reminder may also be sent to the Party concerned towards the end of the six-month period.
- 145. If no response has been received within six months, the secretariat informs the Committee accordingly, and notifies the Party concerned that it has done so.
- 146. In some cases, the Committee may be content to base its deliberations solely upon the information included in the submission and the response thereto. Alternatively, it may decide to use its discretion to gather information from other sources in accordance with paragraph 25 of the annex to decision I/7.
- 147. If prior to commencing its deliberations on the substance of the submission the Committee decides that a hearing will be needed (see paras. 182-189 below), the Party concerned and submitting Party will be informed of their right to be represented at the hearing in accordance with paragraph 32 of the annex to decision I/7. Subject to financial resources, financial support will be provided where needed to assist an eligible government representative from the submitting Party and Party concerned to participate.

# (c) Submissions by Parties concerning own compliance

- 148. A submission may be brought before the Committee by a Party concerning its own compliance with its obligations under the Convention. Such a submission shall be addressed in writing to the secretariat and supported by corroborating information.<sup>58</sup>
- 149. The submission should be in the format described in paragraphs 136-139 above.
- 150. The secretariat informs the Committee of any such submissions that it receives and circulates them to the Committee without delay.
- 151. In some cases, the Committee may be content to base its deliberations solely upon the information included in the submission. Alternatively, it may decide to use its discretion to gather information from other sources in accordance with paragraph 25 of the annex to decision I/7.
- 152. If prior to commencing its deliberations on the substance of the submission the Committee decides that a hearing will be needed (see paras. 182-189 below), the Party will be informed of its right to be represented at the hearing in accordance with paragraph 32 of the annex to decision I/7. Subject to financial resources, financial support will be provided where needed to assist an eligible government representative from the submitting Party to participate.

-

<sup>&</sup>lt;sup>58</sup> Annex to decision I/7, para. 16.

# (d) Referrals by the secretariat

- 153. Under paragraph 17 of the annex to decision I/7, the secretariat may refer a case to the Committee if it has reservations about a Party's compliance with its obligations under the Convention. Such a referral shall be addressed in writing to the Committee and supported by corroborating information.
- 154. When in doubt about the situation in a country, the secretariat may request information from the Party concerned as part of its general work or in preparing the synthesis report according to decision I/8 on reporting.
- 155. The secretariat may consult the Committee before requesting information from a Party in the context of the compliance mechanism, if it considers this to be useful. In some cases, this may result in the Committee requesting the secretariat to seek the information from the Party.
- 156. While the secretariat may become aware of possible non-compliance in various ways other than through consideration of the reports (e.g. correspondence, conversations, newspapers, etc.), formal referrals by the secretariat are based only upon information which is published or transmitted to it in written form.
- 157. If letters from the public concerning possible non-compliance are addressed to the secretariat rather than to the Committee and it is unclear whether or not the letter is intended as a communication under paragraph 18 of the annex to decision I/7, the secretariat clarifies the matter with the correspondent, and, if it transpires that the letter is intended to be a communication, it deals with it in the normal manner for such communications. If it is immediately clear, or is subsequently made clear, that such a letter is not intended as a communication, the secretariat informs the correspondent of the availability of the procedure for consideration of communications from the public, where he or she does not appear to be aware of it and invites him or her to consider the possibility of using that procedure.
- 158. If such correspondents indicate that they do not wish to submit a communication in the sense of paragraph 18 of the annex to decision I/7, the secretariat has various options available to it, including consulting the Committee, seeking corroborating information from other sources or taking no action (e.g. on the grounds that its resources should be allocated to other matters having higher priority, that the information is insufficiently solid, that the alleged non-compliance is not of sufficient gravity, etc.). The secretariat uses its discretion in choosing among these options, taking into account the nature of the particular case. The decision by the secretariat on whether to refer a case to the Committee under paragraph 17 of the annex to decision I/7 is final and not subject to appeal.
- 159. The secretariat may, instead of making a referral, invite a Party to consider making a submission in accordance with paragraph 16 of the annex to decision I/7.
- 160. The secretariat informs the Committee when it has requested information about possible non-compliance from a Party in the context of a referral under the compliance mechanism.

161. If prior to commencing its deliberations on the substance of the referral the Committee decides that a hearing will be needed (see paras. 182-189 below), the Party concerned will be informed of its right to be represented at the hearing in accordance with paragraph 32 of the annex to decision I/7. Subject to financial resources, financial support will be provided where needed to assist an eligible government representative from the Party concerned to participate.

# (e) Requests by the Meeting of the Parties

162. The Meeting of the Parties may request the Compliance Committee to examine compliance issues. For example, the Meeting of the Parties may request the Committee to prepare a report on compliance with or implementation of the provisions of the Convention. It may also ask the Committee to monitor, assess and facilitate the implementation and compliance with the Convention's reporting requirements.

163. Depending on the nature of the request, the Committee's procedure for examining a request of the Meeting of the Parties concerning a Party's compliance may take a similar form to that of the Committee's procedure for reviewing the implementation of decisions of the Meeting of the Parties, including with respect to the Party's reporting deadlines (see paras. 211-223 below).

# VIII. Common elements for all compliance review procedures

## Information gathering

164. Under paragraph 25 of the annex to decision I/7, in order to assist the performance of its functions, the Committee may:

- (a) Request further information on matters under its consideration;
- (b) Undertake, with the consent of any Party concerned, information gathering in the territory of that Party;
- (c) Consider any relevant information submitted to it; and
- (d) Seek the services of experts and advisers as appropriate.

165. The provisions apply to all functions of the Committee described in paragraph 13 of the annex to decision I/7, including the consideration of submissions, referrals and communications. In practice, the Committee may apply the provisions on gathering information in different ways depending on the general or specific character of the compliance issue involved and on its trigger (communication, submission, request or referral).

### **Considerations in information gathering**

166. In considering information gathering, the Committee may evaluate the following elements:

#### (i) Type of information needed

- 167. The missing information may be:
  - (a) Background and contextual information, including texts of legislation/regulations in general of the Party concerned and legislation transposing the Convention, facts related to the particular case, such as dates, exact text of a decision, etc.
  - (b) Views and opinions, e.g. on how national legislation works in practice, the underlying reasons for specific problems, etc;
  - (c) Advice, for instance, on how to solve a continuous problem with the application of existing legislation, and on how to influence the practices of members of the public or public administration, etc.

#### (ii) Possible sources of the required information

168. Depending on the type of the missing information, the sources may vary and may include:

(a) Requests to the government of the Party concerned, usually through the national focal point and the relevant public authorities, or to the communicant;

- (b) Requests to NGOs and scientific communities, and academia;
- (c) Literature and other research and analytical material;
- (d) The public;
- (e) The secretariat.

## (iii) Time and cost implications of information gathering

169. In principle, the Committee's approach to information gathering is pragmatic and costeffective, and aims to facilitate the efficient and effective discharge of its tasks. This also
means that the Committee may avoid being overloaded with too much information, and may
only seek additional information when it deems it necessary for the consideration of a specific
matter. If the required information can be made easily available by the Committee members
or the secretariat, there may be no need to request such information from the Party concerned
or the communicant.

#### **Means of information gathering**

- 170. The means of information gathering include the following three groups:
  - (a) Easily accessible and no-cost or low-cost means of obtaining information, including, but not limited to: questions to the Party concerned, communicant or submitting party for and requests for additional information; reports from the Parties submitted in accordance with decision I/8; Committee members; literature; internet, including information made available through the Aarhus Clearinghouse.
  - (b) Obtaining information by contacting external sources, which, depending on the circumstances, may require a decision by the Committee to do so, such as seeking information/opinions/advice from national and international experts, from governments, academia, private sector and non-governmental organizations as well as international organizations working in the Party concerned. A list of experts who have agreed to provide information to the Committee may be established on the basis of experience of the Committee and input from the Committee members.
  - (c) Costly and more complicated means, which require a specific decision by the Committee to do so, for instance invitations to experts to meetings of the Committee and visits by Committee members and/or the secretariat to carry out on-the-spot information gathering and appraisals.
- 171. If one of the parties to a case considers it requires additional time to prepare its reply to the Committee's request for additional information, it may ask for additional time. The Committee generally accommodates requests for at least some additional time and, for fairness, the timeframes for the other parties to the case will be similarly extended.
- 172. The Committee may determine at any stage in the process that further information should be gathered, including in the period before the Party concerned has provided its response.

- 173. In order to avoid last minute provision of information, the Committee may impose a deadline (generally, two weeks before the meeting) by which information to be considered at a particular meeting must be supplied.
- 174. If one of the parties to a case considers it requires additional time to prepare its reply to the Committee's request for additional information, it may ask for additional time. The Committee generally accommodates requests for at least some additional time and, for fairness, the timeframes for the other parties to the case will be similarly extended.

#### **On-the-spot information gathering**

- 175. According to paragraph 25 of the annex to decision I/7, the Committee may undertake, with the consent of the Party concerned, information gathering in the territory of that Party to assist in the performance of its functions.
- 176. On-the-spot information gathering may be undertaken by the Committee or the secretariat and may be facilitated by international/ sub-regional organizations (such as OSCE and UNDP) that are present in the territory of the Party concerned and familiar with the Convention. The Committee will ensure that persons mandated to undertake on-the-spot information gathering understand that they act on behalf and under the instruction of the Committee and that the mission is non-confrontational, non-judicial and consultative in line with the principles laid down in article 15 of the Convention.
- 177. A mission for on-the-spot information gathering is undertaken only if the Committee has consulted with and received the consent of the Party concerned. The Committee may decide to undertake such a mission if it deems it necessary for the consideration of the matter and the information required cannot be obtained through other means. The Committee will consider the following criteria, namely whether:
  - (a) The Committee has enough information already to open a file and the situation of alleged non-compliance is and continues to be serious;
  - (b) The Committee lacks essential information or the case presents serious uncertainties or difficulties as to the appropriate measures that should be recommended; and
  - (c) It is not possible to obtain the missing elements by other less costly means.

# Comments and information submitted during the proceeding

178. The length limit for communications, submissions and responses (see paras. 128 and 136 above and annex 1 below) applies also, as a maximum, to any information or comments submitted by a party or observer in the course of the proceeding. Accordingly, any comments or information submitted should be no more than 6,000 words (ten A4 pages), and generally considerably shorter. In the case of replies to questions from the Committee only, if more than ten pages are required, in no circumstances should the reply be longer than 12,000 words

(twenty A4 pages).

- 179. The paragraphs of comments should be numbered. If any supporting documents are annexed, a list of annexes should be included at the end of the reply or comments. The comments should clearly specify which paragraph of the reply each annexed document relates to. The relevant parts of each annex should be highlighted.
- 180. The submission of extraneous, superfluous or bulky documentation should be avoided. Only documentation essential to the comments should be submitted, including:
  - (a) Relevant national legislation, with the most relevant provisions highlighted.
  - (b) Relevant court decisions and the results of other review procedures, with the most relevant sections highlighted.
  - (c) Other documentation which the party or observer considers directly relevant to the comment or reply, with the most relevant sections highlighted.
- 181. All supporting documentation should be submitted in the original language, together with a legal standard English translation thereof, or if that is not possible, a legal standard translation in either Russian or French.

## Where needed, a hearing with the parties concerned

- 182. Under paragraph 24 of the annex to decision I/7, the Committee may hold hearings. It also has the discretion to commence its deliberations on a case without holding a hearing.
- 183. The purpose of a hearing is to provide clarity on legal and factual issues. In certain cases, the factual and legal issues are already fully clear from the documentation and a hearing does not provide any significant new information not already contained in the documents.
- 184. Accordingly, in each case, the Committee will examine the documentation before it with a view to deciding whether a hearing is in fact needed. In deciding in each case whether a hearing should be held, the Committee will consider the following criteria:
  - (a) Whether there is no, or very limited, disagreement between the parties on the facts of the case; and
  - (b) Whether the underlying legal issues are well defined.
- 185. If after applying the two above criteria the Committee decides a hearing is needed, both parties will be invited and expected to attend the hearing in Geneva in person.
- 186. If on the basis of the above two criteria, the Committee's preliminary view is that a hearing will not be needed in order for it to commence its deliberations, the Committee will invite the views of the parties to the case on whether they consider a hearing would be needed. Observers will also be free to submit their comments on this point. If either party or any observer considers a hearing is indeed necessary, that party/observer should provide a brief explanation (maximum one page) to the Committee of the reason(s) why, in the light of particular factual or legal aspects of the case, it considers a hearing should be held.

- 187. The Committee will thereafter consider the explanation provided by the party/observer before deciding whether a hearing should be held. In considering the reasons provided, the Committee will take a careful approach. Accordingly, if the Committee considers that either a party or an observer provides a substantive reason showing why a hearing is needed, the Committee will hold a hearing.
- 188. When the Committee has decided to hold a hearing to discuss the substance of a case at a particular meeting, the secretariat notifies the parties that the hearing will be held and invites them to participate in the hearing in accordance with paragraph 32 of the annex to decision I/7.
- 189. If the Committee holds a hearing in a case, it is expected that both the Party concerned and the communicant or submitting Party will participate in the hearing in person. Each party should ensure that its representative(s) taking part in the hearing have the necessary competence to answer the Committee's questions within the scope of the case. In certain cases, this may include the participation in the hearing of other relevant ministries or local authorities on the part of the Party concerned.
- 190. Subject to resources, financial support will be provided where needed to assist a representative of the communicant and, for eligible countries, a representative from the Party concerned and, if applicable, the submitting Party, to participate.
- 191. Where it considers it appropriate, the Committee may send questions or a list of issues to the parties in advance of the hearing in order to assist the parties to focus their submissions. In addition, both parties are encouraged to provide their opening statements in writing to the secretariat in advance of the hearing.
- 192. At the hearing, the Party concerned and the submitting Party or the communicant will each be invited to make statements regarding the non-compliance alleged. The Committee will also put questions to the parties on matters that, despite the written information received, do not appear clear to the Committee. This may involve factual as well as legal matters.
- 193. The discussion of any communication, submission, request or referral generally takes the following form:
  - (a) Introduction and opening of the discussion by the Chair;
  - (b) Short presentation of the case by the curator;
  - (c) Opening statements by the communicant, submitting Party or secretariat (if a referral), and by the Party concerned;
  - (d) Questions from the Committee, responses from the Party concerned and the communicant or the submitting Party,
  - (e) Comments from observers at the invitation of the Chair;
  - (f) Final comments by the communicant or submitting Party;
  - (g) Final comments by the Party concerned.
  - (h) Closure with information from the Chair and/or curator of any further request for information to be provided by the parties.

194. It is not permitted to film or record during hearings.

# **Preparation of draft findings**

195. When the Committee has received the response from the Party concerned to a submission,<sup>59</sup> communication, request or referral, or if no response is received when the final deadline for receiving such a response has passed, it will:

- (a) Consider whether sufficient information is available for it to be able to consider the substance of the case, and if not, identify what further information is needed;
- (b) If sufficient information is available, consider whether a hearing is necessary and, if so, schedule a hearing to discuss the substance of the case in open session, or decide to commence its deliberations without a hearing (see paras. 182-189 above);
- (c) After the hearing, if held, or otherwise when the Committee considers it has sufficient information before it, prepare draft findings, measures and recommendations in closed session. In doing so, it may at any time request additional information from the parties or other sources; and
- (d) Finalize its draft findings and, if applicable, recommendations and send them to the parties for their comments;
- (e) Adopt its findings and, if applicable, recommendations taking account of any comments received from the Party concerned, communicant or submitting Party and observers.

196. The Committee will start the preparation of its draft findings and, where applicable, recommendations as soon as it considers that it has a sufficiently complete picture of the case. In preparing its draft findings and recommendations, the Committee will rely on information provided by the parties and observers only after having made sure that both parties have had a fair opportunity to comment on and give their views on the information provided. The Committee will take into account all elements of the case, including the cause and frequency of any non-compliance.

197. In accordance with paragraph 33 of the annex to decision I/7, the Committee prepares its draft findings and recommendations in closed session.

198. If the Committee finds that the Party concerned is not in compliance, it then considers and agrees upon possible measures or recommendations to address the non-compliance found. "Measures" in the sense of paragraphs 33 and 34 of the annex to decision I/7 are understood to refer to measures that the Committee is entitled to take under paragraph 36 (a) and (b) of the annex to decision I/7 pending consideration by the Meeting of the Parties. Of these measures, it is the Committee's standard practice at the time of sending its draft findings to the parties for their comments to invite a Party concerned found to be in non-compliance to

45

<sup>&</sup>lt;sup>59</sup> References to submissions in this section should generally be understood to refer to submissions made by a Party about another Party's compliance, in accordance with paragraph 15 of the annex to decision I/7, rather than submissions by a Party about its own compliance.

agree that the Committee may make its recommendations to it directly in accordance with paragraph 36(b) of the annex to decision I/7 (see also para. 204 below).

- 199. Once prepared and agreed by the Committee, the draft findings with any measures or recommendations are transmitted to the Party concerned and the communicant or submitting Party with an invitation for them to comment within six weeks. It is the secretariat's practice to notify the parties to a case by email before the draft findings are sent.
- 200. Draft findings and, if applicable, recommendations drawn up by the Committee are posted on the website once they have been transmitted to the parties concerned. All comments on the draft findings should be submitted through the secretariat. When submitting comments to the secretariat for the attention of the Committee, parties are encouraged to copy the other party.
- 201. Comments on the draft findings and recommendations should not include information that could have been provided at an earlier stage of the process; such information will not be able to be taken into account. Moreover, the Committee will generally not be in a position to examine legislative or other developments in the Party concerned brought to the Committee's attention after the draft findings have been sent to the parties for their comments. Rather, the Committee will examine any relevant developments in its review of the implementation of the findings and recommendations and any related decision of the Meeting of the Parties on compliance. If in a particular case the Committee exceptionally considers that it would be necessary to examine the legislative or other developments before adopting its findings in the case, the Committee will prepare revised draft findings taking those developments into account. The revised draft findings will be sent to the parties for their comment and thereafter posted on the website.

# **Adoption of findings and recommendations**

- 202. After having taken the comments on the draft findings received by the set deadline into account, the Committee will review, finalize and adopt the draft findings and, if applicable, its measures or recommendations.
- 203. The final version will be prepared as an official document available in the three official languages of the Convention and transmitted to the parties concerned. An advance unedited copy of the adopted findings is uploaded on the Committee's website shortly after the findings have been communicated to the parties. Until the production of the document as an official United Nations document, editorial or minor substantive changes (that is changes which are not part of the editorial process and aim at correcting errors in the argumentation but have no impact on the findings and conclusions) may take place.

# IX. Follow-up if non-compliance found

## Recommendations directly to the Party concerned

204. If the Committee adopts findings of non-compliance, with a view to addressing compliance issues without delay pending the next session of the Meeting of the Parties it may, subject to the agreement of the Party concerned, make recommendations to the Party concerned directly in accordance with paragraph 36(b) of the annex to decision I/7. This is to provide an opportunity for the Party concerned to address the problems identified in advance of the next session of the Meeting of the Parties.

205. The Committee will thereafter invite the Party concerned to provide progress reports on its implementation of the Committee's findings and recommendations. Communicants, submitting Parties and observers will be given the opportunity to comment on the Party's progress to implement the Committee's finding and recommendations.

206. After taking into account the information received, the Committee will prepare a report to the next session of the Meeting of the Parties on the Party's progress to implement the Committee's findings and recommendations during the intersessional period.

207. If, in its report to the next session of the Meeting of the Parties, the Committee finds that the Party concerned has fully met the Committee's recommendations, then the Meeting of the Parties will be invited to endorse the Committee's findings and to welcome the Party concerned's actions to come into compliance. That should then be the end of the matter. In contrast, if the Committee, in its report to the next session of the Meeting of the Parties finds that the Party concerned has not yet fulfilled the Committee's recommendations, then the Bureau to the Convention<sup>60</sup> in preparation for the upcoming session of the Meeting of the Parties, will prepare a draft decision concerning the compliance of the Party concerned based on the findings and recommendations of the Committee and will thereafter submit the draft decision to the Meeting of the Parties for its consideration and possible adoption (see paras. 209-210 below).

208. If the Party concerned does not agree that the Committee may make recommendations to it in accordance with paragraph 36(b) of the annex to decision I/7 or if only a few months remain before the next session of the Meeting of the Parties, the Committee will instead transmit its recommendations to the Meeting of the Parties directly (see paras. 209-210 below).

# **Consideration by the Meeting of the Parties**

209. Prior to each session of the Meeting of the Parties, the Committee transmits each of its findings and, where applicable, recommendations adopted during the intersessional period to

<sup>60</sup> See http://www.unece.org/env/pp/bureau.html.

the Meeting of the Parties for its endorsement. Regarding the Parties found to be in non-compliance, in preparation for the upcoming session of the Meeting of the Parties, the Bureau to the Convention<sup>61</sup> prepares a draft decision concerning the compliance of each Party concerned based on the findings and recommendations of the Committee and submits the draft decision to the Meeting of the Parties for its consideration and possible adoption.

210. The Meeting of the Parties will make the final decision on specific measures aimed at bringing about full compliance with the Convention. The decisions of the Meeting of the Parties are communicated directly to the parties and made public. In its decision on compliance, the Meeting of the Parties will request the Party concerned to provide reports to the Committee at regular intervals (generally once per year) during the next intersessional period on its progress to implement the measures set out in the decision. During the intersessional period, the Committee will examine the progress made by the Party concerned and report to the next session of the Meeting of the Parties on the extent to which the Party concerned has fully implemented the measures set out in the decision (see paras. 211-219 below).

# Committee's review of the implementation of decisions of the Meeting of the Parties on compliance

- 211. In order to effectively examine the follow-up to decisions of the Meeting of the Parties concerning compliance by individual Parties, the Committee prepares periodic progress reviews which examine the extent to which the Party concerned has by that date fulfilled the recommendations set out in the decision of the Meeting of the Parties concerning its compliance. Prior to preparing each progress review, the Committee will invite any communicant or Party that has submitted a communication or submission giving rise to the decision, and any observers who have registered their interest with the secretariat to take part in the follow-up on the decision, to comment on the progress of the Party concerned. This may include comments on the most recent progress report submitted by the Party concerned or the progress to that date made by the Party concerned to implement the decision more generally. The Committee will take into account all information received within the scope of the decision in the preparation and finalization of its progress review.
- 212. The progress review may include advice to the Party concerned on the further actions it may wish to take in order to demonstrate that it has fully met the measures set out in the decision of the Meeting of the Parties concerning its compliance. Once finalized, the Committee's progress reviews are sent to the Party concerned, communicants, submitting Parties and any registered observers and posted on the website.
- 213. The Party concerned, communicants, submitting Parties and registered observers will thereafter be invited to take part in an open session at a meeting of the Committee to discuss the progress made by the Party concerned and the findings and recommendations contained in the Committee's progress review.

\_

<sup>&</sup>lt;sup>61</sup> See http://www.unece.org/env/pp/bureau.html.

214. At the meeting, the Party concerned, communicants, submitting Parties and observers will be invited to participate in the discussion and to provide information on the progress made and the remaining challenges in implementing the decision of the Meeting of the Parties. Where possible, the Committee will use audio or video conferencing to facilitate the participation of Parties, communicants and observers. The Committee will take into account any comments received on the progress review, and any findings and recommendations contained therein, in the preparation of its next progress review or, if the intersessional period is coming to an end, its final report to the Meeting of the Parties

215. When appropriate, the Committee may seek the services of experts and advisers, as set out in paragraph 25(d) of the annex to decision I/7, to assist it in its review of the implementation of decisions of the Meeting of the Parties concerning the compliance of individual Parties.

# Report to the Meeting of the Parties on the implementation of its decision on compliance

216. In advance of each session of the Meeting of the Parties, the Committee prepares its final report on the progress by the Party concerned during the intersessional period to implement the decision of the Meeting of the Parties concerning its compliance. The report will set out the Committee's findings on the extent to which the Party concerned has met each of the measures set out in the decision. The Committee's final report builds upon its progress reviews during the intersessional period, taking into account the further progress made by the Party concerned since the Committee's most recent progress review and the comments received from the submitting Party, communicants and registered observers. If the Committee finds that the Party concerned remains in non-compliance, the Committee's report will also include recommendations to the Meeting of the Parties on how that non-compliance should be addressed. The Party concerned, as well as communicants, submitting Parties and observers, are given the opportunity to comment on the Committee's draft report to the Meeting of the Parties before it is finalized. Once finalized, the report is sent to the Party concerned, communicants or submitting Parties and observers, transmitted to the Meeting of the Parties for its consideration and posted on the website.

217. In line with paragraph 207 above if, in its report to the next session of the Meeting of the Parties, the Committee finds that the Party concerned has fully met the Committee's recommendations, then the Meeting of the Parties will be invited to endorse the Committee's findings and to welcome the Party concerned's actions to come into compliance. That should then be the end of the matter. In contrast, if the Committee, in its report to the next session of the Meeting of the Parties finds that the Party concerned has not yet fulfilled the Committee's recommendations, then the Bureau to the Convention<sup>62</sup> in preparation for the upcoming session of the Meeting of the Parties, will prepare a draft decision concerning the compliance of the Party concerned based on the findings and recommendations of the Committee and will thereafter submit the draft decision to the Meeting of the Parties for its consideration and possible adoption (see paras. 209-210 above).

\_

<sup>&</sup>lt;sup>62</sup> See http://www.unece.org/env/pp/bureau.html.

- 218. The Committee makes its findings with respect to its follow-up on decisions of the Meeting of the Parties on the basis of the evidence before it. If that evidence shows that the requirements of the Convention have been met, in the absence of any evidence to the contrary, the Committee should find the Party to be no longer in non-compliance.
- 219. The Committee has however repeatedly made clear that a finding that a Party has fulfilled the requirements of a decision of the Meeting of the Parties in no way precludes the Committee from again examining that Party's legislation or practice concerning the same issue in a future case if relevant evidence is put before it to show that that legislation does not in fact meet all the requirements of the Convention or is not being applied in practice in accordance with the requirements of the Convention.

# Any developments subsequent to the finalization of the Committee's report to the Meeting of the Parties

- 220. It is the responsibility of the Party concerned, communicants, submitting Parties and observers to inform the Committee in good faith of any relevant developments in the law or practice of the Party concerned.
- 221. If any new information is submitted after the Committee has finalized its report to the upcoming session of the Meeting of the Parties on the implementation by the Party concerned of the decision concerning its compliance, either:
  - (a) If a further decision concerning that Party's compliance is to be adopted by the Meeting of the Parties at its upcoming session, and the new information relates to the remaining points of non-compliance within the scope of that decision, the Committee will examine the new information provided in the course of its follow-up during the next intersessional period on the new decision of the Meeting of the Parties; or
  - (b) If there is no decision concerning that Party's compliance to be adopted by the Meeting of the Parties at its upcoming session or the new information relates to matters outside the scope of the new decision, the new information may be submitted in the form of a communication (by a member of the public) or submission (by a Party).
- 222. In addition, the Meeting of the Parties may request the Committee to act pursuant to paragraphs 13 (b) and (c) of the annex to decision I/7 (see paras. 162-163 above).

#### Issuance of a caution

223. In exceptional circumstances, the Compliance Committee may, after taking into account for example the attitude and lack of serious engagement of a Party concerned to address its outstanding non-compliance, recommend that the Meeting of the Parties issue a caution to a

Party concerned under paragraph 37(f) of the annex to decision I/7. Since its first findings were reported to the second session of the Meeting of the Parties (Almaty, 2005), the Committee has recommended that the Meeting of the Parties issue a caution to four Parties.

# X. Preparing a communication – useful information

# Summary of key points<sup>63</sup>

- Only Parties have legal obligations under the Convention, and therefore issues of compliance can arise only with respect to Parties. Accordingly, Signatories and other States which are not Parties to the Convention fall outside the competence of the Compliance Committee.
- A State's legal obligations under the Convention commence from the moment it became a Party. Except by way of background information, communications should address only actions, omissions, events or situations which occurred when the State in question had legal obligations under the Convention, i.e. after it became a Party.
- In considering any communication from the public, the Compliance Committee will take into account the extent to which any domestic remedies (i.e. domestic review or appeals processes) are available to address the issues raised in the communication, except where such a remedy would be unreasonably prolonged or obviously would not provide an effective and sufficient means of redress. In every case, before making a communication to the Committee, the member of the public should explore whether the problem could be resolved by using any such domestic remedies.
- Communications to the Committee may concern either a general failure by a Party to introduce the necessary legislative, regulatory and other measures to implement the Convention; specific deficiencies in the measures taken; or specific instances of a person's rights under the Convention being violated; or a combination of these. For communications concerning a person's rights under the Convention, it must be stressed that the compliance procedure is designed to improve compliance with the Convention and is not a redress procedure for violations of individual rights.
- The compliance mechanism aims to facilitate compliance by Parties with their obligations under the Convention. It is not intended as a redress mechanism. The mechanism itself and any measures undertaken in the course of, or as a result of, the Compliance Committee's operation are by their nature non-confrontational, non-judicial and consultative.

52

conort bands of text in the previous parts of the Guide.

<sup>&</sup>lt;sup>63</sup> This section builds on text in the previous parts of the Guide.

## Who can submit a communication

224. Any member of the public (i.e. any natural or legal person), may submit a communication to the Committee. The person submitting a communication (the communicant) does not have to be a citizen of the Party concerned, or, in the case of an organization, to be based in the Party concerned.<sup>64</sup> For the same reason, the communicant does not need to be a citizen of, or be based in, a Party to the Convention.

225. The communication should provide basic information – name and contact details – on the identity of the communicant, whether this is an individual or an organization. If the communicant is a registered organization, the communication should be signed by a person legally authorized to sign for the organization and the name and contact details of that person should be provided. If the communication is made by a group of persons, a contact person should be designated to correspond on behalf of the group and the personal information provided for that person. The Committee will not consider anonymous communications.

226. It is not necessary that the communicant be represented by a lawyer or that the communication is prepared with legal assistance. However if some legal knowledge is available to the communicant, this may improve the quality of the communication and thus facilitate the work of the Committee. In cases where a communicant is represented by someone else (e.g. a lawyer or other representative), the communicant is required to confirm in writing to the Committee that it has authorized this person to represent it in connection with the communication in question. The Committee may request a full power or other document to show that the person duly represents the organization or group of persons.

# Specify the Party concerned by the communication

227. The communication should clearly identify the Party to the Convention (the "Party concerned") whose compliance is the subject of the communication. Where a person wishes the Committee to examine alleged non-compliance by more than one Party, a separate communication should generally be submitted for each Party concerned. The Committee may decide that related communications be managed jointly.

228. A communication may be made concerning any Party to the Convention, provided that:

- (a) The Convention is in force for that Party. The Convention enters into force for a State on the ninetieth day after the date on which it has deposited its instrument of ratification, acceptance, approval or accession.<sup>65</sup>
- (b) The Party has not "opted out" of the compliance mechanism with respect to

<sup>&</sup>lt;sup>64</sup> Unless the context indicates otherwise, the term "State" is understood to also cover any regional economic integration organization that is entitled to become a Party to the Convention under its article 19, such as the European Union.

<sup>&</sup>lt;sup>65</sup> The list of States that have ratified, accepted, approved or acceded to the Convention can be found at http://www.unece.org/env/pp/ratification.html.

communications from members of the public.<sup>66</sup> (While this opportunity exists, to date no Party has opted out).

# Timing of a communication and of the related facts

229. Communications may be made one year or more after the date of the entry into force of the Convention for that Party. In other words, during the first year after the entry into force of the Convention for a Party, there is a grace period during which the Committee may not consider communications from members of the public with respect to that Party.

Example: State X deposits its instrument of ratification on 1 July 2019. The Convention enters into force for that State 90 days later, i.e. on 28 September 2019. Communications may be made with respect to that Party from 28 September 2020.

- 230. Importantly, this does not mean that the Convention is not binding for the Party during the one-year grace period; after the one-year grace period is over, communications may be submitted concerning events which occurred during the first year after the entry into force of the Convention in the Party concerned.
- 231. If the significant events giving rise to a communication occurred before the entry into force of the Convention for that Party, the Committee is likely to determine that it will not consider the communication, as the State had no legal obligations under the Convention at the time of the events.
- 232. Communications should be submitted no later than six weeks before the meeting at which they may be forwarded to the Committee for a determination of preliminary admissibility (the dates of all upcoming meetings are listed on the Committee's website). Communications received after the six-week cut-off date may be considered for preliminary admissibility at the following meeting.
- 233. The Committee considers all admissible communications, but it may decide to consider communications in a different order than the order they have been received, on the basis of the need for adequate review of compliance by the Parties and the Committee's workload.

<sup>&</sup>lt;sup>66</sup> When a Party has "opted out", it means that it has notified the Secretary-General of the United Nations that it does not accept the consideration of communications from the public for a period of up to four years, as allowed for in decision I/7 (annex, para. 18). Such a notification should have been made before 23 October 2003 for States that were Parties at the time when the decision was adopted; for other States, the notification should be made no later than one year after the entry into force of the Convention for that State. If a Party has made such a notification, it is not possible to make a communication concerning that Party for a period of five years after the entry into force of the Convention for that State or such shorter period as may be specified in the notification to the Depositary.

## Form of the communication

- 234. A communication must be in writing and should be in the format for communications provided in annex I to this Guide. The format for communications is also available on the Committee's website.<sup>67</sup>
- 235. Communications should be as concise as possible. The communicant should avoid including any information that is not essential to establishing the existence and nature of the alleged non-compliance.
- 236. If the secretariat receives information from a member of the public which purports to be a communication to the Committee, but which is not in the format of a communication or which does not refer to and clearly does not concern compliance with the Convention, the secretariat, if necessary after consulting with the Chair and Vice-Chair(s), will inform the member of the public that the information cannot be treated as a communication and explain the requirements for communications.

# Presentation of the facts of alleged non-compliance

237. The communication should set out, in chronological order, the facts on which the communication is based.

# The nature of alleged non-compliance

- 238. For each of the provisions with which the communicant alleges the Party concerned has not complied, the communication should clearly explain how the Party concerned has breached that provision based on the facts of the case. The communication should contain all the information that would be needed to establish the alleged non-compliance.
- 239. Any key supporting documentation that will help to substantiate the communicant's allegations should be attached to the communication.
- 240. A communication may concern:
  - (a) A general failure by a Party to take the necessary legislative, regulatory or other (e.g. institutional, budgetary) measures necessary to implement the Convention as required under article 3(1);
  - (b) A failure of specific legislation, regulations or other measures implementing the Convention to meet specific requirements of its provisions;
  - (c) Specific events, acts, omissions or situations that demonstrate a failure by the public authorities of the Party concerned to comply with or enforce the Convention.

\_

<sup>67</sup> https://www.unece.org/env/pp/cc/com.html.

# Provisions of the Convention relating to the alleged noncompliance

241. The communication should clearly list the specific provisions (articles, paragraphs and sub-paragraphs) of the Convention, which the communicant alleges that the Party concerned has failed to comply with.

## Use of domestic remedies

242. As explained in paragraphs 116-120 above, when determining whether the communication is preliminarily admissible, the Committee will take into account the extent to which the available domestic remedies have been exhausted before the case was brought to the Committee. The communication thus should clearly specify which, if any, steps have been taken to use domestic remedies. A failure to provide this information may result in the communication being found inadmissible.

243. If no domestic remedies have been used or if there are other domestic remedies still available, the communication must explain why they have not been used, for example because no remedies were available or because they were prohibitively expensive or unreasonably prolonged. If it is claimed that the domestic remedies are either too expensive or prolonged, the communication should provide sufficient evidence to show the typical cost or timeframe for such cases. If remedies were sought in connection with the matter which is the subject of the communication by a person other than the communicant, or by another person in a closely related case, this should also be stated in the communication.

# Use of other international procedures

244. The communication should include information on whether the subject matter has been submitted to other international procedures, including the steps taken, when they were taken and what the results were.

# **Confidentiality**

245. If the communicant is concerned that the disclosure of information submitted to the Committee could result in his or her being penalized, persecuted or harassed, he or she is entitled to request that such information, including any information relating to his or her identity, be kept confidential. The same applies if the communicant is concerned that the disclosure of information submitted to the Committee could result in the penalization, persecution or harassment of any other person.

- 246. Subject to paragraph 249 below, in the absence of a clear request for confidentiality, no information submitted to the Committee will be treated as confidential.
- 247. If the communicant requests any information to be kept confidential, the communication should clearly highlight all information for which confidentiality is claimed, and also provide a second version of the communication with the confidential information redacted. The communicant should also provide a short explanation of why confidentiality is claimed.
- 248. While the Committee respects requests for confidentiality, it is important that there is enough information in the communication for the Committee to examine the case, and for the Party concerned to understand the case it should answer. In some cases, if the request for confidentiality relates to a large volume of information or to information that is necessary to substantiate the allegations in the communication, this may impede the Committee's consideration of the case. Finally, if the communicant requests that his or her identity be kept confidential, it is strongly recommended that he or she indicates a representative, such as a lawyer or NGO. Therefore, while there is no restriction in requesting confidentiality, this right should be exercised only when it is considered absolutely necessary.
- 249. On some rare occasions, the secretariat or the Committee may consider that the communication contains information that may place another person or persons at risk of being penalized, persecuted or harassed. In such circumstances, after informing the Chair, the secretariat will ask the communicant to prepare a redacted copy of the communication or supporting documentation for forwarding to the Party concerned and for posting on the Committee's website.

# **Supporting documentation**

- 250. Supporting document should be kept to the minimum necessary to substantiate the allegations made in the communication. Such documentation may include:
  - (a) Relevant national legislation, with the most relevant provisions highlighted.
  - (b) Relevant decisions/results of other review procedures, with the most relevant sections highlighted. Any judgments or decisions of other review bodies in support of the arguments of the communicant or of the Party concerned should be dated after the Convention entered into force for the Party concerned.
  - (c) Relevant correspondence with the public authorities of the Party concerned or other documentation that substantiates the alleged non-compliance, with the most relevant sections highlighted.
- 251. All supporting documentation should be provided as annexes. For fairness and due process, the Committee will not take into account information provided through hyperlinks (see further para. 41 above).
- 252. A list of attached annexes should be set out at the end of the communication and numbered in Arabic numerals. There should be one document per annex. The communication should make clear to which paragraph or sentence of the text each annex relates.

## Language of the communication and related documentation

- 253. Communications should be submitted in one of the official languages of the Convention, i.e. English, French or Russian. If a communication is submitted in Russian or French, the secretariat will arrange for its translation into English. The communicant may comment on the accuracy of the translation if he or she so wishes. Supporting documentation will also be translated as the Committee considers necessary and subject to available resources.
- 254. As English is the working language of the Committee, if a communication or other documentation is not submitted in English, this may considerably slow down the process of its consideration.
- 255. In order for the Committee to consider documents in languages other than the official Convention languages, the communicant should provide a translation, preferably in English and submit it together with the original. Certified translations are preferable to unofficial. In some situations the Party concerned may be asked by the Committee to provide the translation, particularly of legislation.
- 256. The need for translation of any supporting documentation submitted in languages other than the official ones will be considered on an ad hoc basis.

#### To whom communications should be addressed and how

- 257. Communications should be addressed to the Committee but sent to the secretariat at the address indicated at the end of annex I of this Guide.
- 258. The communication should be sent by email, with the enclosures attached. If the total file size of the communication and attachments is more than 10 megabytes, then the files should be sent via a free file transfer software, such as wetransfer.com.
- 259. The receipt of a communication will generally be swiftly acknowledged by the secretariat. If the communicant does not receive acknowledgement of receipt within one week, he/she is encouraged to send an email to check if the communication was safely received. If the communicant does not receive a response to this further email within the next week, he/she is encouraged to send his/her communication in hardcopy by registered mail. This is to make sure that communications are not lost due to any technical difficulties that may occur in their transmission.
- 260. The communication and other correspondence relating to a case should not be sent to the individual members of the Committee or to its Chair; the secretariat will forward the communication and related correspondence to the Committee.

# **Annex 1: Format for communications**

#### **Important note:**

The communication should be no more than 6,000 words (ten A4 pages). If in an exceptionally complex case more than ten pages are required, in no circumstances should the communication be longer than 12,000 words (twenty A4 pages). The paragraphs of the communication should be numbered and a list of annexes provided at the end.

#### I. Information on correspondent submitting the communication

1. [Full name of organization or person(s) submitting the communication

Permanent address

Address for correspondence on this matter, if different from permanent address

Telephone

**Email** 

If the communication is made by a group of persons, provide the above information for each person and indicate one contact person.

If the communication is submitted by an organization, provide the following information for the contact person authorized to represent the organization in connection with this communication:

Name

Title/Position

Telephone

Email]

#### II. Party concerned

2. [Name of the Party concerned by the communication]

#### III. Facts of the communication

3. [Detail the facts and circumstances of the alleged non-compliance. Include all matters of relevance to the assessment and consideration of your communication. Explain how you consider that the facts and circumstances described represent a lack of compliance with the provisions the Convention.]

#### IV. Provisions of the Convention with which non-compliance is alleged

4. [List as precisely as possible the provisions (articles, paragraphs, subparagraphs) of the Convention that you allege the Party concerned has not complied with]

#### V. Nature of alleged non-compliance

- 5. [For each of the provisions with which you allege non-compliance, clearly explain how you consider that the Party concerned has failed to comply with that provision based on the facts of your case. (Provide as attachments to your communication the key supporting documentation that will help to substantiate your allegations).]
- 6. [Also indicate whether the communication concerns a specific case of a person's rights of access to information, public participation or access to justice being violated as a result of the alleged non-compliance of the Party concerned, or whether it relates to a general failure by the Party concerned to implement, or to implement correctly, the provisions of the Convention. If you consider that the non-compliance concerns a general failure by the Party concerned, provide as attachments to your communication any key supporting documentation that will help to substantiate that it is a general failure.]

#### VI. Use of domestic remedies

- 7. [Describe which, if any, domestic remedies have been invoked to address the particular matter of non-compliance which is the subject of the communication. Specify which domestic remedies were used, when they were used, what claims were made, what the results were and whether there are any other domestic remedies available.]
- 8. [If no domestic remedies have been invoked or if there are other domestic remedies available, explain why they have not been used. This information will be important for the Compliance Committee's decision on admissibility of the case.]

#### VII. Use of other international procedures

9. [Indicate if any international procedures besides the Aarhus Convention Compliance Committee have been invoked to address the issue of non-compliance which is the subject of the communication. If so, specify which procedures were used, when they were used, what claims were made and what the results were.]

#### VIII. Confidentiality

10. [Note that unless you expressly request it, none of the information contained in your communication will be kept confidential. If you are concerned that you may be penalized, harassed or persecuted, you may request that information contained in your communication, including information on your identity, be kept confidential. If you request any information to be kept confidential, clearly indicate which information. It is also helpful for the Committee to know why confidentiality is requested.]

#### IX. Supporting documentation (copies, not originals)

- 11. [Insert a list of annexes at the end of your communication. Clearly specify in your communication which paragraph of your communication each annexed document relates to. For each annexed document, highlight those parts which are essential to your case.]
- 12. [Avoid including extraneous, superfluous or bulky documentation. Attach only documentation essential to your case, including:
- Relevant national legislation, highlighting the most relevant provisions.
- Relevant decisions/results of other review procedures, highlighting the most relevant sections.
- Relevant correspondence with public authorities of the Party concerned or other documentation that substantiates your allegations of non-compliance, highlighting the most relevant sections.]
- 13. [Provide all supporting documentation in the original language, together with a legal standard English translation thereof, or if that is not possible, a legal standard translation in either Russian or French.]

#### X. Signature

14. [Sign and date the communication. If the communication is submitted by an organization, a person authorized to sign on behalf of that organization must sign it.]

#### **XI.** Sending the communication

15. Send the communication by email to the Secretary to the Compliance Committee at the following address:

#### aarhus.compliance@un.org

16. In the exceptional case that you do not receive an acknowledgement of receipt from the secretariat by email within one week, send the communication by registered post to the following address:

Secretary to the Aarhus Convention Compliance Committee
United Nations Economic Commission for Europe
Environment Division
Palais des Nations
CH-1211 Geneva 10, Switzerland

\_\_\_\_\_

# Guide to the Aarhus Convention Compliance Committee

This Guide explains the functions of the Aarhus Convention Compliance Committee and its working methods for Parties, communicants and observers that may engage with the Committee's procedures.

Convention on Access to Information, Public Participation in Decision-making and Access to Justice in Environmental Matters (Aarhus Convention)

Environment Division
United Nations Economic Commission for Europe

Palais des Nations CH-1211 Geneva 10, Switzerland Phone: +41 22 917 4226

Email: aarhus.compliance@un.org

Website: http://www.unece.org/env/pp/welcome.html